Relph, Tiffany Louise (2016) Categorising Interaction between Teachers and Students in ESOL Classrooms. Masters thesis, York St John University.

Downloaded from: https://ray.yorksj.ac.uk/id/eprint/2565/

Research at York St John (RaY) is an institutional repository. It supports the principles of open access by making the research outputs of the University available in digital form. Copyright of the items stored in RaY reside with the authors and/or other copyright owners. Users may access full text items free of charge, and may download a copy for private study or non-commercial research. For further reuse terms, see licence terms governing individual outputs. Institutional Repository Policy Statement

RaY

Research at the University of York St John For more information please contact RaY at <u>ray@yorksj.ac.uk</u>

Categorising Interaction between Teachers and Students in ESOL

Classrooms

Tiffany Louise Relph

'Submitted in accordance with the requirements for the degree of Masters by Research'

York St John University

Department of Languages and Linguistics

October 2016

The candidate confirms that the work submitted is their own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material. Any reuse must comply with the Copyright, Designs and Patents Act 1988 and any licence under which this copy is released.

© 2016 York St John University and Tiffany Louise Relph.

The right of Tiffany Louise Relph to be identified as Author of this work has been asserted by her in accordance with the Copyright, Designs and Patents Act 1988.

Acknowledgements

Course Y: To Emma, Roisin, Kate, Simon and everyone at YSJ International for their help and support in setting up the Course Y classes. To Kim Lofthouse and the team at the York St. John Business School Admin Office, for lending us cameras and dictaphones and for supervising our textbooks! To Katie Evans, Jennifer Bramley and especially Rachel Dry (for lending me the use of her print account!), my fellow Course Y committee members for putting up with my midnight organisational meltdowns. To all the ELT Project students 2014/2015 for sacrificing their own time and teaching expertise, free of charge: we couldn't have made this work without all your help! A big special thanks to all of our hardworking English-learners who attended our free Course Y classes and gave their permission to be filmed for our corpus.

Dissertation: Special thanks to my supervisors Helen Sauntson and Indu Meddegama for their invaluable support, guidance, advice, and book loans (which I haven't forgotten about, Helen!) and to Becky Muradas-Taylor for acting as a sounding board and helping to shape most of the structure of this thesis.

To the Idiots Assembled for accompanying me on all of my research trips to York and to Tony Donaldson and the rest of the Shoebox Zoo, who have both helped and hindered me in equal measure, but I am thankful for all of it! To my cat Louis, who always tried his hardest to prevent me from working by lying across my laptop keyboard- sorry I had to lock you in the kitchen!

And I suppose the most obvious person I have to thank here is N.A. Flanders, without whose work on classroom interaction, this paper could never have been written.

Abstract: The purpose of the present study was to investigate how the Flanders Interaction Analysis Category System (FIACS) can be used to investigate the interactional practices of students and teachers of English to speakers of other languages (ESOL). Additionally, this study sought to produce a system by which interaction could be categorised in the ESOL classroom which could be deployed by teachers in order to help them gain a deeper insight into the ways that they interact with their students during classes. The dissertation tracked the practices of 46 individuals.

The data of this study included recordings of ESOL lessons. These recordings were then transcribed and categorised into different interaction types using a revised version of the Flanders Interaction Analysis Category System, in order to provide insight into what types of interaction most commonly occur between teachers and students and the purpose of that interaction.

The results of this research found that students in the ESOL classroom are given greater encouragement and opportunity to speak during classes than in non-ESOL classrooms, and the classes themselves are more student-centred. Previous research conducted in a non-ESOL context showed a more teacher-centred approach. The results of this study indicate that students and teachers in an ESOL class may have different interactional preferences for learning than those of non-ESOL teachers and students and in order to account for these differences, a new and more ESOL-specific category system has been developed.

Table of Contents

CHAPTER 1: INTRODUCTION	8
1.1 General Introduction to the Study	9
1.2 Purpose of the Study	11
1.3 Research Questions	14
1.4 Significance of the Study	15
1.5 Thesis Outline	16
1.6 Conclusion	17
CHAPTER 2: LITERATURE REVIEW	19
2.1 Introduction	19
2.2 Foreign Language Interaction	19
Sociocultural Theory	20
Interactionist Theory	22
Input Hypothesis	23
Output Hypothesis	24
2.3 Interaction in Education	25
Definition of Interaction	25
Interaction in ESOL	26
Interaction Partner	27
Student-Teacher Interaction	27
Interaction Purpose	28
2.4 Motivation	29
2.5 Communication and Interaction	30
2.6 Quality of Interaction	30
2.7 Introducing FIACS	33

The Rule of Two Thirds: A Teacher-Centred Approach	38
2.8 The Purpose of this Study	40
2.9 Conclusion	42
CHAPTER 3: METHODOLOGY	
3.1 Introduction	43
3.2 Research Questions	43
3.3 Research Design	44
Setting	44
Participants	45
Data Collection	46
Teachers	47
Students	48
Learning Tasks	49
3.4 Ethical Considerations	50
3.5 Coding of the Data	50
3.6 Evaluation of the Original FIACS (Relph, 2015)	61
3.7 Recommendations for Revisions to FIACS (Relph, 2015)	63
3.8 Data Analysis Procedure	65
3.9 Conclusion	66
CHAPTER 4: RESULTS (RESEARCH QUESTION ONE)	
4.1 Introduction	67
4.2 Demographic Information	67
Course Participants	67
Participant demographics	69
Student Gender	70

Student Age	70
Student Nationality	70
Previous Language Learning Experience	71
Teacher Demographic Information	71
Summary: Demographic Information	72
4.3 Research Question One	72
Revising the FIACS	73
Evaluation of the revised Category System	83
Advantages and Disadvantages of Using Interaction Analysis C	Coding
Systems in the ESOL Classroom	85
4.4 Conclusion	87
CHAPTER 5: RESULTS (RESEARCH QUESTION TWO)	
5.1 Introduction	89
5.2 Research Question Two	89
Interactional Purpose	90
5.3 Observing Classrooms Using the Revised FIACS	90
5.4 Conclusion	109
CHAPTER 6: RESULTS (RESEARCH QUESTION THREE)	111
6.1 Introduction	111
6.2 Practical Application of Interaction Analysis Coding Systems i	n
Teacher-Training	116
Interaction Analysis in Micro-Teaching	118
6.3 Conclusion	119
CHAPTER 7: CONCLUSIONS	120
7.1 Introduction	120

7.2 Pedagogical Implications	120
Interactional Partner	120
Interactional Purpose	121
Language Learners	124
7.3 General Recommendations for ESOL Courses	125
7.4 Limitations of the Study	128
Course Participants	128
Study Design	129
7.5 Recommendations for Future Research	130
7.6 Conclusion	133
REFERENCES	135
APPENDICES	
I Sample Teacher Consent Form for ESOL Course	152
II Sample Student Consent Form for ESOL Course	153
III Sample Consent Form for Involvement in this Study	156
IV Sample Transcript with Coding	158

List of tables

3.1 Flanders Interaction Analysis Category System (1970)	53
4.1 Group A Participant Demographics	69
4.2 Group B Participant Demographics	69
4.3 Teacher Demographics	71
4.4 Adapted and Modified FIACS	81
5.1 Percentages of Interaction Categories	91

CHAPTER 1: INTRODUCTION

Over the past few years there has been a significant interest in the ways in which teachers and students interact with each other in the classroom, as well as evaluative techniques used in teacher training throughout different educational levels and in the interactional management of groups of students such as in classes and meetings. Changing views on the pedagogy of teaching have focused attention quite markedly on the interactional side of teaching behaviours.

Yet, even without considering the great power and value it would have to take the opportunity to assess interaction, focusing on this direction, the limitation of conducting observation-led research, does generally, remain considerable. The use of interaction analysis category systems, whilst alleviating certain problems, also generates others which may only be fully realised through the study of existing literature on, for example, skills and information theory, person perception, and interaction analysis. Very little of the current literature in these fields has been effectively applied to the understanding of the process of interaction between teachers and students in ESOL classrooms; though if reliable and objective styles of analysing teacher-student interaction are to be developed in order to, for example, improve teachers' self-awareness of their teaching style, the need for a more precise understanding of these factors is critical.

This project is concerned with interaction between teachers and students in ESOL classrooms (English to Speakers of Other Languages). With these ideas in mind, it is interesting to consider how contexts such as ESOL might be capable of revealing important information about how teachers and students interact with each other in class

and provide a vast number of ways for different interaction to take place. With the inherent social nature and variety of the interactions available, ESOL is perhaps better understood as the breeding ground for rich social interaction. It is therefore imperative that such mediums are considered when it comes to classroom discourse and interaction research.

1.2 General Introduction to the Study

This study focuses on an ESOL course, known as Course Y for the purpose of this study. To investigate how the students and teachers interact verbally, this study tracks students and teachers participating in an English Language course. By examining data on communicative tasks embedded in a course, we can better understand how teachers and students interact with each other in lessons and their preferences for interaction and interaction types, as the needs and interests of ESOL students may differ significantly from the needs of non-ESOL students. This research will expand on current knowledge of ESOL classes as a means to model the teaching and interaction strategies present in this instructional setting.

The demand to learn English is now a force so strong that it is driving the growth in ESOL education. Teaching itself has transitioned from a highly individualised form of instruction, to encouraging students to collaborate with both their classmates and the teacher (Benati, 2013). Consequently, it would seem like the natural progression in ESOL research would be to examine the collaborative environment which students are encouraged to participate in.

This study began life as part of the wider Course Y project. The Course Y project is an initiative setup by the researcher with a multi purpose mission. It provides free English Language tuition to non-native English speakers living and working in York. The second purpose was to provide academics an opportunity to collect data for research purposes. The third purpose was to provide newly qualified teachers an opportunity to volunteer to increase their teaching experience and boost their CVs.

Carrying out classroom research in authentic contexts is important in order to provide accurate results. However, gaining access to these authentic contexts can prove challenging for researchers. Out of this need to obtain authentic classroom data came the Course Y project: a classroom set up to provide data for researchers by recording all of the lessons for observational purposes and also giving researchers the opportunity to attend the sessions and gain access to the students as potential research participants.

In order to store the recordings made from the lessons securely and to allow researchers to gain access to the dataset whilst preserving the personal details of the participants, the Helix Media Library was used. Video and audio recordings of the lessons were uploaded to the York St John Helix Media Library under the category Course Y. The recordings were placed behind a password which meant that only people with an account could view them. This prevented the data from being made public, as anyone wanting to view it would need to obtain permission from the Course Y committee who have the power to grant access. The files are encrypted which prevent them from being shared outside of the Helix Media Library.

Despite its steady growth, ESOL has been associated with significant challenges when

it comes to interaction between teachers and students. As interest for learning increases, so does the need to develop programmes that meet the needs of a new student population in a foreign language (FL) environment. FL learning creates a dually challenging environment in which interaction becomes vital, both for language development and learning outcomes, but also because it aids in a strong sense of community, and limits isolation, which are essential factors for student persistence.

1.3 Purpose of the Study

Many teacher-training programmes have been devised and implemented to improve the efficiencies and competencies of Primary and Secondary school teachers, with the majority of educational research focusing on children as learners. However, the reverse is true in the context of adult education, especially in the ESOL classroom. Sharan Shahi (2010) stated that there is a great need for more research to be conducted in the field of teachers' interaction patterns at this level to help inform future teacher efficiency, and also to discover what kind of teachers' behaviour best contributes to students' language acquisition (Yang, 2003).

Flanders (1970:24) speaks of the "sense of urgency" about obtaining "authentic behaviour" in research. Contradictions abound about the way we think of education and what actually goes on in the classroom. Today, most university students in the UK are prepared for teaching and teachers in schools undergo in-service education without specifically being taught how to identify, to practice producing and analyse teaching behaviour. Concepts for thinking about teaching are frequently used, though they may not be able to be defined in terms of specific actions which teachers can perform. Perhaps another issue is that the universally accepted notion of 'one year teaching experience' is a standard unit for setting salaries and achieving status as a teacher. This creates little incentive for creating a program of self-development in which more than 'one year of experience' can occur in one year. To improve the quality of teaching, the pattern of interaction events must first be examined and judgments made about their quality. Jackson (1990) believes that even before learning teaching methodology, teachers should learn the principles of establishing relationships well and how to apply these principles to their lessons.

There have been many tools and systems devised to study classroom interaction patterns. Among these the 'Flanders Interaction Analysis Category System' (FIACS) has been found to be the best known and most widely used. The purpose of interaction analysis according to Flanders (1970: 3) is to "study teaching behaviour by keeping track of selected events that occur during classroom interaction. Other events are to be taken into consideration only when this would assist the observation." One application of this is to assist teachers and trainee teachers in developing and controlling their teaching behaviour. The second application of studying classroom interaction is to identify variations which occur in the chain of classroom events, focusing on teaching behaviour and its relationship to classroom interaction. Flanders claimed that this system was applicable to all classroom contexts. In 2015, I attempted to address this claim by examining his original theory in the context of an ESOL classroom. Relph (2015) found that the original FIACS failed to provide an accurate reflection of interactional practices in ESOL-specific environments due to its lack of accommodation for ESOL-specific features of interaction. That study recommended a revised version of FIACS be drawn up to be employed in exclusively ESOL contexts in order to provide an accurate picture of what constitutes significant interactional behaviour in this context and which also could help teachers and teacher trainers evaluate and assess teacher performance in the classroom. This present study sets out to fulfil those recommendations.

This study has two primary purposes. The first is descriptive: to provide a detailed understanding of how students and teachers interact in ESOL classes, in order to better understand how they communicate with each other. Many previous studies believe that interaction is crucial for FL learning (Hatch, 1978; Long, 1980; Swain, 1995). However, there has been little exploration of actual interactional practices in ESOL, as the majority of the current research in this learning context relies on self-assessment measures.

In this thesis, the relevant literature is reviewed with special reference made to the particular value of categorising interaction in ESOL.

This study concerns an environment in which students take an active role in the lessons which are most beneficial for their language development, as well as determining how many lessons they will participate in. In most academic environments learners are given specific guidelines as to how they need to interact on the course. Instructor assessment is often based on the frequency of interaction. However, this study offers the opportunity to understand actual learner practices, independent of instructor expectations.

The second goal of this study is to revise and update the Flanders Interaction Analysis

Category System (FIACS) in order to enable it to provide an accurate way of categorising the interactions in an ESOL context, which could help teachers to self assess their own teaching styles and provide insight into how they interact with their students in class.

1.4 Research Questions

This project is an exploratory study seeking to develop an in-depth picture of how EFL learners interact in a classroom environment. Additionally, it will develop a system by which to categorise these interactional practices by seeking answers to the following research questions:

Research Question 1: Based on the recommendations outlined in Relph (2015), is the revised and updated FIACS an appropriate way to categorise ESOL classroom interaction between teachers and students?

The first research question is a conceptual one. It examines the changes made to the FIACS in the revised and updated version and questions whether or not these changes have improved the accuracy of the system in regards to ESOL-specific classroom contexts.

Research Question 2: What does an analysis of student-teacher interaction, using the modified Flanders Interaction Analysis Category System, reveal about the different interaction types produced by the teachers and students in the context of an ESOL classroom?

The second research question is an analytical one. This question examines interactional practices of learners in numerous ways, including the total number of interaction types occurring, the percentage of direct versus indirect talk, interactional partner communicated with, the interactional purpose, the collaborative nature of the task, and the various media used.

Research Question 3: How could an interaction analysis category system be used to inform the teacher training of ESOL teachers?

This research question examines ways in which the revised FIACS can have a practical application for teachers, teacher trainers and educational bodies in terms of monitoring and evaluating teaching technique with the assumption that interaction between teachers and students in the ESOL classroom is linked to better student grades, retention and course satisfaction.

1.5 Significance of The Study

The findings from this study have both theoretical and pedagogical significance. Firstly, this study adds to the field of classroom interaction, specifically how teachers and students interact with each other in an ESOL context. Additionally, this study adds to the recent growing body of research on interaction in ESOL by providing a framework with which interaction can be categorised in order to show exactly the types of interaction taking place and by whom. Taken together, this study sheds light on a growing body of students, namely ESOL learners.

1.6 Thesis Outline

This chapter has provided a general overview of the Masters thesis, specifically in regard to interaction in ESOL. The next chapter offers a review of the previous research that is used as a theoretical foundation for this study, with specific attention paid to interaction analysis category systems, including FIACS. Chapter Three contains a detailed description of the methods of inquiry used in this study, including the research design, measures, data collection and analysis procedures. From here, the results are divided into addressing each of the three research questions in individual chapters 4-6. Chapter Four addresses Research Question One, Chapter Five covers the discussions raised in Research Question Two and Chapter Six presents the findings of Research Question Three. Finally, Chapter Seven presents a summary and discussion of the findings, including the outcomes of the research, the limitations of the study, and a discussion of the pedagogical implications of the findings, including recommendations for encouraging interaction in ESOL classes and recommendations for future research.

This project furthers existing knowledge in the areas of ESOL and classroom interaction. Firstly, it cannot be ignored that a wide range of academic literature has been published in relation to classroom interaction, and ESOL. However, what this project does is to build upon such works by investigating the topic through an alternative context.

Secondly, this project takes ideas and concepts developed within the field of classroom

interaction and investigates if, and how, they are applicable to the landscape to ESOL teaching, allowing for insight to be given into classroom discourse on this growing interaction platform.

The third, and arguably most exciting contribution to knowledge this project will make, is through devising an interaction analysis category system to be applicable in an ESOL context. In this way, we will be able to see what is significant interactional behaviour in ESOL classroom interaction. By recording interaction in this context, this will provide a rare insight into how teachers and students interact with each other in a real-life ESOL classroom context.

Not only will this study fill a gap in academic knowledge, but it will also addresses issues outside of academia, for instance teacher-training. Research into interaction in ESOL classes will be beneficial to the language-learning industry and enable them to communicate with and target their communication more effectively. An example of how discursive research can be used within a variety of institutions and businesses can be seen within Stokoe's CARM project, which offers communication skills training that can be adapted to any workplace (See Stokoe, 2011). Stokoe's project has proved that discursive research, such as in this project, can help businesses improve interaction, services and therefore financial return.

1.7 Conclusion

This project aims to provide a unique insight into the world of ESOL classroom

interaction, through capturing live, real-time interaction in ESOL between teachers and students in classrooms. As the demand for ESOL grows and EFL teaching becomes a growing industry, it seems imperative that we understand the interaction that occurs on such mediums so that as teachers we can tailor our interactions to best suit the needs and preferences of the students. This thesis aims to build on existing knowledge by:

- 1) Applying an alternative analytical lens (that of discourse analysis).
- Applying knowledge of previous interaction analysis studies to a new ESOL-specific context.
- Capturing real life classroom communication live through the use of recording technologies.
- 4) Producing a framework with which teachers and teacher trainers can use to record their own interactions in the classroom, possibly using this information to monitor their teaching approach.

CHAPTER 2: LITERATURE REVIEW

2.1 Introduction

This chapter presents a summary of the relevant literature on teacher-student interaction. Interactionist theories in foreign languages (Hatch, 1978; Long, 1980) posit that learning is enhanced through social interaction. In addition to interaction, increased input (Krashen, 1985) and output (Swain, 1995) have also been recognised as essential factors in learning outcomes. Studies have shown that interactive tasks within lessons may lead to a higher rate of progress in the class (Northrup, 2002; Tello, 2002; Restauri, 2006).

Since the 1970s, an increase in academic interest has been seen in relation to interaction practices and classroom teaching. Academic interest in this area has risen as it has become more apparent to scholars that classroom interaction, especially that which occurs between a teacher and student, can play an integral part in the learning development of the student. Key ideas about the classification of interaction may need re-examining in the context of the field of English to speakers of other languages (ESOL).

2.2 Foreign Language Interaction

The present study investigates interaction in ESOL classrooms where the students' first language (L1) is not English. Because there is little agreement in any one theory, I will begin with an overview of the most relevant Second Language Acquisition (SLA) theories for this study: Sociocultural theory (Vygotsky, 1978; Hall, 1993; Lantolf, 2000; Swain, 2000), Interactionist theories (Hatch, 1978; Long, 1983; Pica, 1987), the Input Hypothesis (Krashen, 1985), and the Output Hypothesis (Swain, 1985). Each of

these theories were developed out of a growing discontent in the field of foreign language education which emphasised rote grammar instruction, resulting in the competence of grammatical structures, but not in communicative acts. Each of these theories recognises the vital role communication plays in foreign language learning. However, each has its own perspective as to what the most fundamental aspect of language learning is.

Interactionist theories have highlighted the importance of Vygotsky's sociocultural, constructivist beliefs essentially stating that language is rule-governed, and learned through interacting with others (Hatch, 1978; Long, 1983; Pica, 1987). By contrast, the Input Hypothesis (Krashen, 1985), which attributes the language learning process as an innate skill, and the Output Hypothesis (Swain, 1985) have focused on nativist approaches to explaining the language process.

Sociocultural Theory

According to Vygotsky's (1978) Sociocultural theory, all learning is situated in interaction or mediation between humans. A theory originally developed in educational psychology, this theory is the foundation from which a number of SLA theorists built an understanding of how language is learned. In brief terms, his theory states that development is a combination of interaction with people and the tools that the culture provides to help form their own view of the world.

The Sociocultural theory posits that learning cannot be separated from the context in which it takes place, as Vygotsky (1978: 57) explains:

Every function in the child's cultural development appears twice: first, on the social level and, later on, on the individual level; first, between people (interpsychological) and then inside the child (intrapsychological). This applies equally to voluntary attention, to logical memory, and to the formation of concepts. All the higher functions originate as actual relationships between individuals.

At the core of this theory is the notion that learning cannot be done in isolation. Learning must be within a social context.

Vygotsky also distinguishes between the developmental level attained (what a learner can do with assistance), and the level of potential development reached by "problem solving under adult guidance, or in collaboration with more capable peers" (1978: 86). Vygotsky refers to this as the Zone of Proximal Development (ZPD). The ZPD implies that a learner is in the process of mastering certain structures, and this higher level of development can only be achieved through collaboration with others. In order for the ZPD to function properly, two aspects must be included: subjectivity and scaffolding. Subjectivity is a process whereby two individuals arrive at a shared understanding when at first having a different understanding. Scaffolding is when the support of the peer is gradually lessened.

Sociocultural theory, a general learning theory (not specific for learning languages), focuses on the importance of collaboration for cognitive development. It has been widely applied to studies in the field of SLA (e.g., Lantolf, 2000), its application being that language learners advance to higher levels of linguistic competence when they collaborate and interact with speakers who are more knowledgeable than they are.

In summary, many researchers agree that interaction is at the core of language

acquisition. However, Carroll (2001) reminds us that it is still unclear as to what exactly interaction provides, and if it is anything more than practice. Additional uncertainties centre around whether interaction should be viewed as one-way (Input or Output Hypothesis) or two-way (Interactionist theories). Finally, there is still uncertainty as to whether one-way theories or two-way theories have the dominant influence on language acquisition. Regardless of this uncertainty, Chapelle (1997) believes that the majority of researchers of instructed SLA would agree on the value of learner interaction for language development; whether they take a cognitive (Gass, 2003) or, more recently, a sociocultural perspective toward interaction (Lantolf, 2001).

This study is most concerned with foreign language (FL) exchanges within an ESOL course. By a close examination of how students and teachers interact on a course, it is possible to test which of the four theories (Interactionist theories, The Input Hypothesis, The Output Hypothesis, and the Sociocultural theory) is most significant to their learning, and thus further our understanding of SLA theories in the context of an ESOL course. This can be done by a thorough examination of how teachers and students actually interact. Having established a basis from which our understanding FL interaction is built, I will now discuss the background literature of Foreign Language interaction.

Interactionist Theories

Many pedagogical approaches used in classroom learning draw their basic guiding principles from research that has extolled the role of interaction in Second Language Acquisition (SLA). Hatch (1978) changed the direction of SLA research by suggesting that interactions offer more than simply a forum in which to practice specific grammatical forms. She argues that interactions should be examined in the context of their social process, and that language learning evolves out of learning how to converse. So, while much of the research in the field of SLA suggested that a learner should master a form and then practice that form through conversation, Hatch maintained that learners acquire grammatical structure from first learning how to communicate.

A number of other studies developed based on Hatch's insights, the most applicable being Long's (1980) Interactional Hypothesis (de Bot, Lowie & Verspoor, 2005). Long (1980) believed that learning occurs when attempting to communicate in the target language; students learn from the process of interacting. His hypothesis states that negotiation of meaning is what has the greatest impact on acquisition. When negotiating meaning, learners do everything from asking for clarifications, receiving feedback, conducting comprehension checks, shifting topics, and repeating the language. This process draws the learner's attention to the mismatch between the conversational partner's input and the learner's output. To summarise, Long believes the key for successful language development is the process that the learner goes through when they receive input.

Input Hypothesis

In contrast to Interactionists' views of language being a learned skill, nativists, such as Krashen (1985), believe that humans are programmed to take in new language, that learning language is an innate ability. Krashen maintains that language acquisition follows a natural order that learners will eventually master, so as long as learners are always given input that challenges them to the next level, they will understand the meaning of the other language, and eventually will master the forms of the language as well.

In order to accomplish this natural progression, Krashen believes instruction in the language being learnt should mimic first language acquisition and that formal learning settings should mirror that of natural (non-academic) environments (Terrell, 1982). Krashen's theories are outlined in his "Monitor Model" which includes five hypotheses at the core of this theory, summarised here:

First is the acquisition/learning hypothesis. This asserts the separation of acquisition from learning; acquisition is seen as a natural process, where there is no conscious focusing on linguistic forms. Learning, on the other hand, is a conscious, unnatural process, marked by conscious knowledge of the rules of grammar. Secondly, the monitor hypothesis claims that learned knowledge can function only as an editor or monitor for output. Next, the input hypothesis states that the most important factor in the development of second or other language proficiency is the learner's exposure to the target language, which he explains should be language that is one step ahead of the learner's current stage of comprehension. Fourthly, the natural order hypothesis claims that acquisition takes place in a predictable order and is unaffected by instruction. Finally, the affective filter hypothesis provides reason for learners' varying success. This suggests that SLA can only occur when the conditions are optimal (such as the student having self-confidence, motivation, and a low anxiety level). Krashen's Monitor Model highlights the necessity of input. It further suggests that teachers need to ensure that students are given input appropriate for the language level in order to maximise language acquisition.

Output Hypothesis

Also a nativist, but at the other end of the spectrum, Swain (1995) believes that input is not sufficient for the development of language abilities. Swain's Output Hypothesis has

drawn attention to the importance of learners' production of the target language. According to Swain (1995), acquisition is achieved by producing the target language, which increases learner fluency by giving opportunities for students to experiment with the language and get feedback from other learners. Students then notice a 'gap' in what they are able to say and what they want to say. The key for successful language learning, in the framework then, is the explicit attention given to productive language skill (speaking and writing).

More recently, Swain (2000) uses the term "collaborative dialogue" in lieu of "output". Swain believes that dialogue provides learners with an opportunity to use language and to reflect on language use. This adaptation draws on Vygotsky's Sociocultural theory, specifically that originates from interaction with others.

2.3 Interaction in Education

Despite the commonly held theory that communication and interaction between students, and students and teachers, can have a positive influence on the quality of the education (Roblyer & Ekhaml, 2000), there has not been much empirical study exploring classroom interaction in an ESOL context. Although interaction has long been recognised as an essential factor in FL learning outcomes, research on interaction is still emerging as technologies expand our pedagogical capabilities, and our understanding of what interaction can incorporate is still developing.

Definition of Interaction

Interaction research is an ever-expanding field. So much so, that even though there have been a number of studies conducted, there has not been a consensus in the field as to what interaction entails. For the purpose of this study, it is important to understand

how the term interaction will be used.

This study combines Wagner's definition of interaction as "reciprocal events requiring [a minimum of] two objects and two actions. Interactions occur when these objects and events mutually influence one another" (1994: 8). So for example, interaction requires that there be at least two people (in our case a teacher and student(s)) and that these people are both engaging in some sort of verbal action together. The verbal action must be a reciprocal exchange for us to say that interaction has occurred. And because this study is also rooted in SLA theory, it takes into account Vygotsky's (1978) Sociocultural theory that learning is a social process. Thus, this study focuses on the interaction between *humans*, namely those between the teacher and student, and does not venture into the effects of interaction with content or media. While many studies also use the term interactivity, Wagner differentiates interaction (human contribution) from interactivity (technological capabilities).

Interaction in ESOL

ESOL interaction has much in common with interaction in a non-ESOL environment, but because students are dealing with interacting in a second language, the interaction that takes place on ESOL courses may be more challenging than that in a non-ESOL classroom.

Much of the recent research on ESOL interaction highlights the primary role interaction (FL or otherwise) has in determining the quality of the programme, and that interaction positively affects the effectiveness of ESOL courses. ESOL encourages students to collaborate and work together under the direction of an instructor. As such, it would seem that a natural progression in ESOL research is also to examine the interaction that takes place.

Many researchers believe that a higher level of communication and interaction in ESOL can have a positive effect on students' achievement (Zahed-Babelan & Moeni-Kia, 2010; Abdolrahimi, Zahed- Babelan & Namvar, 2013; Odundo, 2013). However, Barnes & Todd (1995) argue that students are not given enough credit to take care of their own learning independent of interaction from the teacher.

Interaction Partner

Three types of interaction were identified by Moore & Kearsley (1996): (1) Student-Teacher Interaction (e.g., a student working with the teacher), (2) Student-Student Interaction (e.g. a student working with another student or a group of students), and (3) Student-Content interaction (e.g., a student working through the course materials). However, this model assumes that there is always a one-teacher classroom. It does not account for interaction that may involve more than one teacher in the same classroom at the same time interacting with the students. So Moore & Kearsley's model cannot be said to encompass *all* types of classroom interaction.

Student-Teacher Interaction

It is not surprising that several studies have found significant value in communication between student and teacher. Gutierrez (2000) explains that the role, and interactions with, the teacher are an essential part of the learning experience.

Research on student-teacher interaction has revealed that students perceive interactions between themselves and their teachers as important to their participation in class, and to the quality of the teaching and their own learning. Jiang & Ting's (1998) study reported that the quality and quantity of teacher interactions positively correlated to perceived learning (Jiang & Ting, 1998), and that students who felt they knew their instructor participated more actively in group discussions (Thurmond et al., 2002).

Interaction Purpose

Although the most widely discussed area of interaction is described in terms of the partner with whom one is interacting, an increasing number of studies have investigated the purpose of interactions. A number of studies have investigated the role of productive and receptive interaction, but Hao's (2004) is the only study that measures student attitudes towards four types of interactive tasks: (1) instructional (anything considered to be content centred), (2) affective (social), (3) collaborative (productive), and (4) vicarious (receptive). Her study found that type of interaction the students liked the most was instructional interaction, and that they liked vicarious interaction the least. But because the study was based on perceptions of interaction, not practices, it is not certain which types of tasks students engaged in the most.

Contradictory to what one might expect, research has yet to show that completion of fewer productive tasks tends to lead to decreased learner outcomes. In fact, just the opposite might be true. Haalen & Miller's (1994) study found that more interactivity (defined in their study by the number of phone calls students made) did not positively correlate to greater learner outcomes. Their data showed that students who interacted very infrequently and those who interacted very frequently had a lower gain score than those who were somewhere in the middle. Similarly, Beaudoin (2002) found that students who are highly interactive do not necessarily get better grades. Beaudoin (2002) explains this point further:

In considering how learning occurs [...] it may be assumed that learning correlates closely to what is visible (e.g., student's written words that appear on the page). It may also be concluded that if there is no visible activity, then little or no learning is likely to occur. [...] Instructors note that some students are passive and non-participatory, sometimes to the point of not even appearing for classes. Yet, despite doubts about how much these students are learning and how well they will do on assignments or exams, many of these same students eventually manage to do quite well academically, regardless of their lack of active face-to-face participation.

2.4 Motivation

Many studies have investigated the link between motivation and language learning. Meunier (1998) addressed the role of situational motivation (including anxiety and comfort, risk-taking, sociability and teaching styles) and task motivation. These two types of motivation (Brown, 1994) are an expansion of FL motivational theory first suggested and widely accepted by Gardner & Lambert (1972), namely the distinction between instrumental and integrative. Meunier's study of third-year French and German students found that participants were intrinsically and socially motivated by discussions.

Based on frequently cited attrition rates (Carr & Ledwith, 2000; Carr, 2000) it would seem that at the beginning of a course there is a very high level of motivation (both situational and task) and that students are not always able to harness this motivation throughout the duration of the course. This is certainly connected to Lambert (1991) emphasising the importance of offering very short courses, much shorter than typical semester length courses. This helps students reach a goal that seems attainable, so they do not get overwhelmed.

One final note on the FL motivation studies is that the majority of the findings in the

FL classroom have been with advanced level students. In addition, the FL students tended to be older (Lambert, 1991). This could have an effect with comfort level in interaction.

2.5 Communication and Interaction

Communication is defined differently according to which approach is being examined. In data processing theories, communication is considered as a method to minimise doubt, maximise confidence and act as an entry point to a particular situation and state. Based on this Abdolrahimi, Zahed- Babelan & Namvar (2013) define communication as "a flow through which two or more interactive agents are involved with each other so to provide the possibility to exchange messages and codes in a flexible way and thus to achieve a goal" (2013: 1).

Interaction is established through at least two individuals being involved in a series of dynamic and constant events in which each person leaves an effect on and is affected by the other within a framework of a mutual determining system (Abdolrahimi Javid et al, 2013). Therefore, interaction refers to the actions and reactions of the group members towards one another; in other words the members of the group influence each other (Safari, 2010). This study aims to explore, verbal interactions between teachers and students during lessons.

2.6 Quality of Interaction

Past research has found that the quality of teacher/student interactions plays a key part

in the quality of the learning (Needles, 1988; Darling Hammond, LaPointe, Meyerson, & Orr, 2007). Good teachers who can forge and maintain good relationships with students play a vital role in schools (Gibbons 2003; Luk, 2004) and such teachers can change the environment within the classroom, bringing support and success to their students (Christie, 2002). According to Reynolds & Peter (2009), one of the most basic characteristics of a good teacher is the ability to establish interaction in the classroom. In this study, a good teacher is one which has the ability to establish interaction, as the main issue observed in classrooms comes from the lack of interaction. Emphasising this issue, Brower *et al.* (2001) state that teaching happens only when a teacher establishes interactions with the students.

It should be noted that talking in and of itself cannot be considered interaction. Interaction is the way in which action and reaction between individuals is organised and this action and reaction forms part of an individual's social desirability (Gage & Berliner, 1998). Blatt *et al.* (2008) conducted relational analysis on interaction and concluded that 65.2% of interaction is related to speech. In contrast, Putz & Aertselaer (2008) estimate that around 65% of meaning transfer happens through non-verbal behaviour.

Fredericksen et al. (2000) reported that the most significant variable for academic achievement was teacher-student interaction. Another study, Jiang & Ting (2000), has suggested that the quantity and quality of teacher-student interaction is linked to student learning. However, Lie (2008) contradicts these claims and found that there is no relationship between interaction partner and linguistic outcome; language development is not dependent on with whom the student is communicating, just that the student is communicating, practicing, and using the language.

In the area of FL research, Lie's (2008) finding also contradicts the Sociocultural theory (Vygotsky, 1978), which suggests the Zone of Proximal Development is done through interacting with a more linguistically competent person, either that of a teacher or a more competent peer. These results would suggest that the competency level of the interaction partner is not significant.

While the results of these studies do not mirror the results from the previous two studies, two things should be reiterated. Firstly, that the previously mentioned studies were all based on student perceptions of interactions and not on actual practices. Secondly, only certain types of activities were included in the analysis. The teacher did not participate in any of the activities which were analysed, therefore the validity of the measurement of students' interactional presence is questionable.

Interaction between students and teachers within the classroom is clearly important in fostering a rapport which may help students become more motivated to learn (Amidon & Hunter, 1967; Needles, 1988; Cole & Chan, 1994; Darling-Hammond, 2008). Interaction is also an affective, temperamental matter and not just a question of someone saying something to someone else. Without this mutual respect between teacher and student, the building of confidence, and the creating of many opportunities, classrooms will remain quiet places with inhibited students who dare not try to express themselves (Rivers, 2000). Oranu (2010) emphasises that classroom talk should be focused, engaging and relevant; theory should be based on actual experience and should make all implications clear and accurate.

2.7 Introducing FIACS

One of the most basic elements of high quality teaching is the way in which students respond to teaching activities, and the interaction between the teacher and student on the whole (Sahlberg, 2007). One method to carry out, with good reliability and validity in regards to interpretation of results is the Flanders Interaction Analysis Categories System (FIACS) (Wragg, 2002; Hai & Bee, 2006; Saba, 2007 and Cenoz & Hornberger, 2008). The FIACS contains ten categories, the first seven of which record teacher talk, with the next two covering student talk. The last category records silence in the classroom. Thus the FIACS examines only verbal behaviour; because this can be observed with higher reliability than nonverbal behaviour. It is assumed that an individual's verbal behaviour is an adequate reflection of their total behaviour (Amidon & Hunter, 1967). It is also assumed that classroom interaction occurs as a series of events, and that teaching behaviour consists of patterns of behaviours, embedded in the series of classroom events. Often during classroom interaction, the same sequence of events can reoccur again and again; such a sequence can be identified as a pattern if it occurs frequently enough to be of interest. The major feature of the FIACS lies in the analysis of initiation and response which are the characteristics of interaction between two or more individuals. Teaching behaviour has been defined "as acts by the teacher which occur in the context of classroom interaction" (Flanders, 1970: 4). FIACS measures directedness/indirectedness of speech and allows for student participation. Silence or confusion is the only nonverbal category.

According to Medley & Mitzel (1958), the FIACS was one of the most sophisticated interaction analysis categories developed at the time of writing. It was unique in the

presenting of a certain amount of information regarding the sequence of behaviour.

Major criticisms of the FIACS are that it lacks attention to cognitive processes (Lambert *et al.*, 1965; Amidon, 1967; Smith & Othanel, 1968), neglects levels of thinking of students and teacher (Amidon, 1967), neglects unique features of the individual students as it concentrates on the class as a whole (Lambert *et al.*, 1965), uses only overt behaviour categories thus removing the concept of purposiveness of behaviour (Boyd & DeVault, 1966), is difficult to learn the system (Amidon. 1967); and is not sophisticated enough or conceptually complex enough to provide necessary information for useful analysis of teaching (Amidon, 1967).

Criticisms levelled at the categories are that some categories need further subdivision, for example the category of criticism, and that there are not enough student-talk categories (Lambert *et al.*, 1965; Amidon, 1967).

Amidon & Hough (1967) were the first to give training in interaction analysis to teachers. Their research showed that teachers who had been trained in interaction analysis were observed by their superiors as being more effective in their teaching than teachers in the control group.

In another study conducted with teachers at Temple University, Furst (1967) reported that teachers who were taught interaction analysis accept student ideas and student behaviour significantly more than teachers not trained in interaction analysis. In addition, it was found that teachers trained in the system showed evidence of significant positive change in attitude toward teaching during the teaching experience when compared with teachers not so trained. They seemed to be more aware of their own verbal behaviour than the untrained teachers. The verbal behaviour pattern of teachers who had been trained in interaction analysis before teaching showed a mean of 66.7% teacher-talk and a mean of 25.55% student-talk in their lesson presentations. Patterns of those trained in interaction analysis while teaching showed a mean of 61.2% teacher-talk and 39.7% student-talk in lesson presentations.

Over the past twenty years, there has been development in methods of analysis of interaction within peer groups. So much so, that to review them all here would be impossible. Instead, focusing on another system that has produced an alternative approach to the study of interaction is Barnes and Todd (1977, 1995). Their framework is an analytic system for studying peer group talk, designed from the bottom-up, as opposed to the top-down approach derivative of a pre-existing set of categories. As a consequence of this, their system was able to take into consideration the context in which this peer talk was occurring. This system focuses on actual processes of interaction and the ways students developed and constructed knowledge without direct teacher influence. They analysed dialogues that took place amongst groups of students and identified the different types of talk and their impact upon the construction of meaning during group interaction. Taking both a social and a cognitive approach, they developed a system to describe speech acts, operating across two levels focusing on

They identified 'exploratory' speech characteristics such as invitations to modify or surmise, tentativeness in voice intonation, assertions and questions made as hypotheses rather than direct assertions, hesitation and changes of direction, and self-monitoring and reflexivity. Out of this, several conditions for collaborative work in classrooms were proposed, directly based upon the empirical evidence. These conditions were further categorised as: initiating, eliciting, extending and qualifying (Barnes and Todd, 1995).

In spite of there being some limitations in the analytical system and the fact that the data was recorded on tape recorders only (meaning information from non-verbal interaction was not recorded and analysed), the work of Barnes and Todd made an important contribution to the study of student-talk. It brings together ideas from discourse analysis and conversation analysis, with research into learning theory and educational instruction. Other studies have used the Barnes and Todd frameworks to explore classroom interaction (e.g. Edwards, 2005), however because of the lack of focus on teacher-talk, and the emphasis this system puts on group-work to provide accurate results, it was deemed that this system was not the most appropriate for this study, which wanted to examine the interactional relationship between teachers and students.

Conceptually, Barnes and Todd (1995) has made a vital contribution to our increasing understanding of the different types of talk and the different interactional relationships between teachers and students. One of the limitations of this method, though, is that what is being analysed is the group as a whole, so this framework does not account for which individual students are participating in the interaction. Consequently, this framework does not explain how the rules and customs of interaction in that context are actually formed. Additionally, by focusing entirely on student talk, this system can not give a completely accurate picture of interaction in the classroom. The SETT (Self Evaluation of Teacher Talk) framework described in Walsh (2006) was intended to provide a springboard into facilitating discussion of classroom interaction amongst teachers and to enable some sense to be made of the interactional organisation of an L2 classroom. The SETT framework is designed to develop teaching skills via classroom interaction. Walsh states that the L2 classroom is a real social context and that what happens inside it matters just as much as what happens outside (Walsh, 2013). Walsh's SETT framework was designed to assist teachers in gaining a better understanding of the context of classroom teaching and to understand how interaction works. The SETT framework is made up of four classroom modes and thirteen interactures. Classroom discourse is presented as a series of intricate and interrelated micro-texts. Meanings are co-constructed by teachers and students, and learning takes place through the interaction between teachers and students.

Walsh reiterates that the SETT framework is a generic instrument which is meant to represent interaction, rather than provide a comprehensive account of it. It is designed to be used as a tool of reflective practice for developing teachers.

Seedhouse (2004) states that the organisation of interaction changes as the pedagogical aim changes. The structures and practices of classroom learning move between being teacher-directed, to a structure very like, but notably different from, conversation, where the students produce more initiative talk (Mori, 2007; Seedhouse, 2005). While understanding the roles of teachers and students in classroom interaction is still important, these roles are perhaps not as fixed as previous work has suggested.

According to Flanders (1970), the majority of lesson time is spent on the teacher delivering a one-sided speech, called "the rule of two thirds"; in that in the majority of classes two thirds of the time someone is talking, it is the teacher who speaks. Teachers have a tendency to dominate the classroom discourse (Gross, 1993; Atakins & Brown, 2001). The content of teachers' speeches were mainly explaining and delivering the materials (lecturing). Students often shy away from asking and answering questions and generating ideas within the class. Walker (2002) states that when the students do get to speak, that speech is within the restricted framework of asking and answering questions which only requires utterances of a few words at a time. Most research seems to concur with Flanders's assertion that teacher-talk makes up about 70% of the interaction within classrooms (Saba, 2007; Wragg, 2002; Fathi Azar, 2003; Hai & Bee, 2006).

Pontecorvo (1993) insists that few verbal interactions occur during class. Cieniewicz (2007: 5) claims that the pattern of interactions between the teacher and students is never equal and always directed towards a small proportion of the class. Cieniewicz further claims that, unless the students are individually nominated by the teacher, then on average only four students will actually contribute in any given lesson. This teaching pattern of periods of 'lecturing' intercut with interludes of questions and answers is the most prevalent method of interaction seen in classrooms and also appears to be the most resistant to change. Abdolrahimi, Zahed- Babelan and Namvar (2013) declare that this approach needs changing to take into account the participation of the students. Cole and Chan (1994) believe that interaction plays a fundamental role

in a successful lesson. The ability to establish a positive relationship with the students is one of the main skills required of teachers. The studies conducted by Rosenshine (1971) and Dankin & Biddle (1974) showed that there was a positive correlation between the time that teachers spent talking, the way in which they spoke and their students' overall achievement. However, there is also a large positive correlation between clarity of speech, ability to attract the student's interest, organisation of speech, and the achievement of the students.

It appears that most lessons, according to the research by Flanders (1970), consist predominately of teacher-talk. Lecturing and questioning are the most common types of interaction seen in the classroom. Some research has found evidence for a relationship between teacher interaction styles and students' overall achievement. However, FIACS does not accurately account for the participation of the students, with seven categories to label teacher-talk and only two to identify types of student-talk.

Caution must be advised also: Flanders (1970) and many of the studies that immediately preceded it in the seventies and eighties, were not conducted in an ESOL classroom context. Teaching methodology in the 1970s leant on the teacher-centred model for the most part. More recently, there appears to have been a trend in exploring the relationship between interaction and achievement, with the majority of the research showing in fact that it is greater student-talk time which leads to higher academic results (Zahed-Babelan & Moeni-Kia, 2010; Abdolrahimi, Zahed- Babelan & Namvar, 2013; Odundo, 2013), as opposed to the findings of Rosenshine (1971) and Dankin & Biddle (1974). With relevance to ESOL, much of the emphasis in teaching speaking skills in particular is placed on the students becoming active interactants, initiating and formulating their own ideas and responses to tasks. As such, students are actively encouraged to participate more verbally in classes by the setting of activities to encourage the production of talk and conversation. Teachers also are trained to try to resist a teacher-centred approach for these kinds of lessons. More research should be done which takes into account the different requirements of students within ESOL classes and the different teaching methodology this requires, as opposed to the ordinary high school teaching method, in order for us to determine what is significant interactional behaviour in this context, which brings us to the statement of the problem.

2.8 The Purpose of this Study

To better understand the complexities of interaction in ESOL, Sharan-Shahi (2010) called for research isolating specific dimensions of interaction, noting a specific need for studies that provide a clear picture of interactivity as it currently exists. He goes on to explain that the study of interaction patterns of teachers and students in ESOL classes, will probably help to develop a programme of ESOL education.

Flanders' system of one set of categories to be generalisable across all contexts is unlikely to work, given the context-dependent nature of all language. That is not to say that Flanders' interaction analysis category system does not work in some contexts, indeed it works very well in perhaps *the majority* of contexts. However, ESOL is not one of them. In this report, I am proposing an amended category system to be deployed within an ESOL context. ESOL takes such a vastly different pedagogical approach to most high school methods of teaching, as well as encompassing a great deal of diversity within itself. Perhaps where most of the previous attempts at formulating an interaction analysis category system, in the vein of FIACS, have suffered in the past, is their desire to create and devise one system to be applicable to all contexts. Indeed, Flanders (1970) has been, by far, the main player in teacher-student interaction analysis category systems. While other similar systems have been created, they have all sought to provide a definitive template of interaction analysis categories applicable in all situations. Perhaps this is not the solution. Indeed, it may be the case that we cannot have a generalisable system, because all language is very much contextually-based. Surely we must concede that it would be a mistake to attempt to map the conventions of one interactional context onto another interactional context and expect them to perfectly fit into each other and match up absolutely. An ESOL interaction analysis system would be complementary to Flanders, rather than competitive, as the two would operate in different spheres.

An ESOL interaction analysis system would have to address the issue of certain ESOLspecific phenomena that FIACS does not currently accommodate, such as a way to categorise modelling and drilling. Also, a way to distinguish between meaningful silences (i.e. those which exhibit misunderstanding and those which occur when the students are working quietly) would be useful.

One important aspect to achieve (perhaps the most important) would be that of making such a category system realistically employable. A balance must be struck between not constructing too many categories, making the learning of the system unnecessarily time-consuming, and not making the categories too broad (so that there is uncertainty on the part of observer as to within which category they should label a particular interaction).

2.9 Conclusion

Numerous studies have addressed the types and importance of interaction in FL education, but few have actually examined the interaction of FL learning. Because little is known about how these two areas intercept, the majority of the research in this area relies on what is known of learning in non-FL contexts. Many studies have pointed to interaction as the defining characteristic of FL education models, suggesting that there is great potential in creating an environment in which more opportunities are provided to the students for various types of interactions with their teachers.

Previous research has focused primarily on interaction perceptions and not on actual practices. By examining the actual interactional practices of teachers and students in an ESOL context, we will be able to give a more accurate reflection of the interactional landscape of a class.

CHAPTER 3: METHODOLOGY

3.1 Introduction

The purpose of this study is two-fold. The first objective is to create an interaction analysis category system to categorise interaction that takes place between teachers and students in an ESOL classroom, based on a similar system created by Flanders (1970) to categorise ESOL classroom interaction. In order to do this I investigated teachers' and students' conversational practices in 31 ESOL classrooms. The second part of this study is to apply this category analysis system to data from a group of ESOL courses in order to discover what an analysis of ESOL classroom interaction can reveal about the intercommunication practices and preferences of teachers and students in this specific context. Previous research has focused on perceptions of interaction, rather than actual practices (Hao, 2004; Kearsley, 1999). By examining data on actual practices, we can better understand how students and teachers interact with each other, their frequency of interaction and their needs for interacting. This will enable researchers to build up a picture of the reciprocated actional happenings of an ESOL class. An ESOL specific interaction analysis category system would hopefully be designed in such a way as to be able to be utilised by teachers and teacher trainers to monitor and enhance their own teaching practices.

3.2 Research Questions

This study was guided by the following research questions:

1. Based on the recommendations outlined in Relph (2015), is the revised and

updated FIACS an appropriate way to categorise ESOL classroom interaction between teachers and students?

- 2. What does an analysis of student-teacher interaction, using the modified Flanders Interaction Analysis Category System, reveal about the different interaction types produced by the teachers and students in the context of an ESOL classroom?
- 3. How could an interaction analysis category system be practically applied in the training of ESOL teachers?

The results of the study first employed a descriptive design, the second part of the study employed an inferential, correlational design. The data collection comprises observations of ESOL classrooms. This data provided insight into how students interact with their teachers in an ESOL class.

3.3 Research Design

<u>Setting</u>

The setting for this study was an ESOL class run within a University in the north of England but open to the wider local community. For the purposes of this study, the course name and institution name have been anonymised so the course will be referred to as course Y. Prior to this study, course Y had been developed and administered by the researcher over the course of approximately 12 months. The program was specifically designed for language learners who did not seek academic credit for participation on the course. It was designed to meet the needs of part-time learners of

English as an additional language. Students chose to attend as many or as few of the classes as they found beneficial.

Learners were given an eight week course to attend. The courses were run on evenings so as to fit in around students who may have worked during the day and met their needs.

Participants

56 individuals were invited to participated in this study (the criterion to participate in this study will be discussed in the next section). Of these 56 individuals, 49 signed and submitted consent forms. Three of these 49 were ultimately not included in the analyses because they withdrew their consent to have their results be included in the study. Further discussion of the participants will be given in the next chapter. The subject groups for this study consisted of 46 individuals comprising of 14 teachers and 32 students.

Students were divided into groups based on when they were able to participate. Group A started first and Group B started approximately three months later. This was to ensure that the teachers would have ample time during the first week of classes to orient the students to the course programme. Further details on the demographics of the course participants will be explored in the next chapter.

The participants were all adult learners of English as an additional language and their teachers. They attended free English language lessons run by the observers over a

period of one year. This amounted to 13 hours and 42 minutes of data to be analysed for this study. It was felt that this was a reasonable sample size to collect within the time available for data-gathering as I had only two semesters to gather data, conduct analyses and write a thesis paper. As all the data had to be transcribed and categorised before it could be analysed, it was felt that more than around 14 hours would take up too much time to prepare and analyse than was available over the limited timeframe of this study.

Data Collection

Two video cameras and four digital voice recorders were positioned around the classroom to record the interaction between the students and the teachers. Every student signed a consent form prior to the start of the lessons to allow themselves to be recorded, and for that data to be used by the observers for academic research purposes. Once consent had been obtained, the video recordings were collated together to form the course Y corpus, an online video library of real-world ESOL classroom data. Access to the full corpus is available through the Helix Media Library at the following link: https://hml.yorksj.ac.uk/Browse/CategoryFolder/3 (A login is required).

According to Simpson and Tuson (2003), some advantages of using observations to collect data are that they enable researchers to gain direct access to social interactions. They also give permanent and systematic records of social interaction through the making of recordings. Observations can be applied to a wide variety of contexts and can be used to address a variety of types research questions.

Teachers

There were 14 teachers used in this study. All of the teachers had studied for a degree in English Language and Linguistics at the same University in the north of England. They had also all obtained a qualification in CELTA no more than 12 months previously. The majority of the teachers had not taught a course since qualifying in the CELTA, but all had previous teaching experience from CELTA, and had taught using the BBC Speakout books on the CELTA course. This textbook was chosen specifically because the activities were interaction focused and would promote interaction between the participants, so that it could be observed in this study. Some teachers had previously an additional 8 weeks teaching experience in EFL teaching at summer camps.

The researcher conducted a survey of a group of 10 ESOL teachers. The teachers were asked questions about their classes, the content they teach, the resources they use, the teachers' perceptions of the kinds of talk occurring in the classes, and what aspects of that talk they would find it useful to reflect on. This was presented in an open discussion based format and the teachers' responses to these discussions proved to be most illuminating. Therefore, it was decided to revise the FIACS based on what teachers said that they needed and wanted most from an interaction analysis category system.

The teachers were then taught how to use the revised FIACS and were given a sample transcript from one of the observed ESOL classes to categorise. The purpose of this was to check the usability and practicality of the system; how easily could it be utilised

by teachers. From this initial trial, some of the categories were further revised based on the recommendations and requests of the sample group of teachers. For example, they suggested a separation of interactive and non-interactive silence was necessary in order to provide a clearer idea of why silence occurs in an ESOL class.

For the third research question, engagement with the teachers was important in order to help develop the revised FIACS into a tool for teacher development of reflective practice. Implementing the FIACS in lessons provides an opportunity for teachers and practitioners to reflect on specific elements of their teaching practice, perhaps realising features of their own and others teaching that may not already be apparent to them. In order to enable this, teachers were given an opportunity to view video recordings of their own lessons and categorise themselves according to the revised FIACS. The aim of this was to allow the teachers to gain a greater insight into theirs and their students' interactional practices within a class. This was followed up with an interview with the teachers in which they were asked to discuss their categorisation of their own lessons, what this revealed to them and whether or not they thought that this knowledge would have any effect on the way they taught in the future.

Students

There were a total of 32 students who participated in this study. Of these 32, 22 were females and 10 were males. The students were from a variety of nationalities and backgrounds, although the majority of them were Chinese. All were over the age of 18 at the time of the study with most being between the ages of 18 and 23. The eldest student was 34. Participants all resided in the North of England at the time of the study,

and all had some previous experience of ESOL education up to at least Elementary (A2) level prior to the start of the study. The participants were living and working in the north of England, either in employment or in education, or attempting to improve their English skills in order to be able to work or study. This tied in with their need to enhance their communication skills in English.

Learning Tasks

Sabry and Baldwin (2003) state that students perceive different interaction types differently. Similarly, following the advice of Twigg (2002), the learning materials provided in this course offer a wide variety of text materials and the individual and collaborative activities needed to provide for differences in interactional preferences and needs. Parker (1999) reiterates this point by stating that a "well designed course that has a focus on on interactivity includes many topics for discussion, feedback from students as well as experts, and finally links to sources of pertinent information" (Parker, 1999:16). Great effort was given to provide participants with a wide variety of activities and useful resources.

Examples of tasks and resources on the course include: a course handbook, language DVDs, audio files, workbooks with quizzes and games, and conversation sessions.

There were a variety of interactive opportunities in which students were encouraged to collaborate with each other and with their teachers. Synchronous tasks, according to Nobuyoshi and Ellis (1993), provide the opportunity of spontaneous conversation and exposure to new linguistic forms.

Each week's lesson was split into four 30 minute teaching slots. Each slot was filled by a different teacher and had a different focus. The tasks were taken from the Speakout curriculum and so were activities typically done in an ESOL course. In order to build interaction, the students were encouraged to work collaboratively.

3.4 Ethical Considerations

Permission to conduct this study was obtained from the Business School Research Ethics Committee at York St John University prior to the commencement of this study. The ethical authorisation code for this study is: 151105.

All individuals meeting the requirements for the course were invited to participate in the study. Students were required to be over the age of 18, reside in the place of study, and have at least an Elementary level of English. The researcher found it necessary to limit the study to participants who had at least some previous English language knowledge in order to better facilitate the interaction between them and the teachers.

Teachers were required to be over the age of 18, have an English language teaching qualification (CELTA), were students enrolled on an English Language and Linguistics undergraduate degree course at the same university the north of England and undertaking the English Language Teaching Research Project module on that course.

3.5 Coding of the Data

This study builds on previous work carried out by the researcher which sought to

challenge the claim made by Flanders (1970) that the FIACS could be used to provide an accurate picture of teacher-student interaction in all educational and classroom contexts. The recommendations from that study, which are explained in full below, revealed that there was a need to adapt the FIACS and create an ESOL-specific interaction analysis category system. But first, it is necessary to introduce the FIACS in its original form, to allow readers to familiarise themselves with the system as it currently stands in order to best highlight both its usefulness and its shortcomings as they relate to this particular context.

When designing the coding system, the researcher created categories that would cover specific types of interaction, before the coding of the data began. The data collected from the course provided recordings of all the classes taking place. The researcher then transcribed each video orthographically and coded the interactions between teachers and students. The interactions were coded using the revised and updated Flanders Interaction Analysis Category System. For an example of a transcript coding, see Appendix IV.

The method used in this study was developed by Flanders (1970) at the University of Minnesota between 1955 and 1960. The category system still has many useful applications although efforts to expand upon it and increase the number of categories have been completed albeit with limited success (Sharan-Shahi, 2010). The FIACS is made up of ten categories: seven of which are used when the teacher is talking, two are used when the student is talking, and the last category is used to indicate silence of confusion. So far as teacher-student interaction is concerned, these three conditions (a) teacher-talk, (b) student-talk, and (c) silence or confusion are thought to be totally inclusive of all possibilities, and since Flanders claimed that any event could be classified, this system therefore should permit coding of interaction events at a constant rate throughout the observation. This is essential in order to examine the proportion of time spent in any of the categories. This claim is further backed by Wragg (2002) and Hai & Bee (2006).

In Flanders' original study, conducted in 1970, an observer would sit in the classroom, in the best position to see and hear all the participants. Today, with the availability of video-recording, the observer records their classifications using pre-recorded lessons. This has the advantage of the observer being able to view a lesson multiple times in order to increase the accuracy of their observations.

The observer records the interaction in a tally. Observation works out to one tally every three seconds, with the observer having to decide which category the events just happened would best apply to. The length of the observation chunk is not significant, but the setting of a steady tempo is: as conclusions depend on consistency, not on speed. The numbers which represent each of the ten code symbols are written down in a sequence which follows the statements being coded. If more than one of the categories was observed within a three second period, then all the categories seen were recorded. The observer did not involve their personal viewpoint.

	Response		
	1. Accepts feeling. Accepts and clarifies an attitude or		
	the feeling tone of a pupil in a nonthreatening manner.		
	Feelings may be positive or negative. Predicting and		
	recalling feelings are included.		
	2. Praises or encourages. Praises or encourages pupil		
	action or behaviour. Jokes that release tension, but not at		
	the expense of another individual; nodding head or		
Teacher Talk	saying "Um hm?" or "go on" are included.		
	3. Accepts or uses ideas of pupils. Clarifying, building,		
	or developing ideas suggested by a pupil. Teacher		
	extensions of pupil ideas are included but as the teacher		
	brings more of their own ideas into play, shift to		
	category five.		
	4. Asks questions. Asking a question about content or		
	procedure, based on teacher ideas, with the intent that a		
	pupil will answer.		
	Initiation		
	5. Lecturing. Giving facts or opinions about content or		
	procedures; expressing their own ideas, giving their own		
	explanation, or citing an authority other than a pupil.		
	6. <i>Giving directions</i> . Directions, commands, or orders to		
	which a pupil is expected to comply.		
	7. Criticising or justifying authority. Statements		

Table 3.1 The categories of the Flanders Interaction Analysis Category System (takenfrom Flanders, 1970).

	intended to change pupil behaviour from an	
	unacceptable to an acceptable pattern; bawling someone	
	out; stating why the teacher is doing what they are	
	doing; extreme self-reference.	
Pupil Talk	8. <i>Pupil-talk—response</i> . Talk by pupils in response to	
	teacher. Teacher initiates the contact or solicits pupil	
	statement or structures the situation. Freedom to express	
	own ideas is limited.	
	9. Pupil-talk—initiation. Talk by pupils which they	
	initiate. Expressing own ideas; initiating of a new topic;	
	freedom to develop opinions and a line of thought, like	
	asking thoughtful questions; going beyond the existing	
	structure.	
10. Silence. No verbal i	nteraction takes place between teachers and students.	

Note that there is no value attached to these numbers. Each number is purely classificatory, to designate a particular communication event, not to pass a judgment on that communication.

This section goes into detail describing each category of interaction used for coding in this study and explains what qualifies an interaction for a particular category.

Category 1

Category 1 consists of teacher statements which accept and clarify an attitude or the feelings of a pupil in a non-threatening manner. The feelings may be positive or

negative. This also includes predicting or recalling feelings. Flanders claims these statements are rare and infrequent. Another reason for the low incidence is the use of a special rule which states that the teacher must literally name or otherwise identify the emotion(s) present before the statement can be classified as "1".

Example:

1) T: "This class seems excited!"

This can be categorised as a "1" because the emotion is cited, the teacher offers no value judgement.

Category 2

Praise and encouragement are statements which carry the value judgement of approval. To look directly at a student and nod your head whilst saying, "Mm-mm"- with a certain inflection- is to communicate to the student that they are producing the correct behaviour and the teacher would like more of the same. The difference between Category 1 and 2, is that in 2 there is an element of praise which is missing in 1. Both categories are used for statements which have overtones of warmth and friendliness, but Category 2 adds teacher approval as well.

Teachers seem to have developed many related superficial verbal habits, such as "Right", "Good", "Okay", and suchlike which are often expressed automatically as soon as the student utterance has ended. This is probably due to the fact that reward is seen as a far better motivator than punishment. Genuine praise can usually be separated from these superficial verbal habits.

Examples:

2) T: Have you got the answer to Question 8?

S: Past Continuous

- T: Yeah. What's the next answer?
- 3) T: Have you got the answer to Question 8?
 - S: Past Continuous
 - T: Well done!

You remembered the grammatical form, didn't you?

Both of these examples begin with a teacher question and a student response. So the first two code symbols will be 4 and 8. In 2), the example could be coded as 4, 8, 2, 4. In 3), because the teacher not only gives praise, but suggests the reason for praising the student's answer, the tempo of recording enters in where interactions are split into 3 second increments, and if the praise takes up longer than three seconds to happen, this can be differentiated by recording 4, 8, 2, 2.

Category 3

Teachers can respond to ideas expressed by a student in the following ways:

- A. They can acknowledge them through repeating them.
- B. They can modify the idea and reshape it in their own words.
- C. They can apply the idea and use it to infer the next stage of an analysis of a problem.
- D. They can compare the idea to one expressed by another student or by the teacher.

E. Or they can summarise what was said by the student.

In the FIACS all of these different teacher meanings are encapsulated in Category 3. However, in some of these different responses, it is highly likely that more elapsed time would be required, thus because more time is required, two or mored coded "3's" would be used.

There is some evidence to show that greater incidence of Category 3 is associated with above average classroom measures of both student satisfaction and student achievement (Flanders, 1970).

Examples:

4) S: I would want to live with Sarah.

T: You would want Sarah as your flatmate. Why is that?

5) S: Because she is a girl.

T: You would prefer to live with a girl.

In 4) the teacher acknowledges the student's answer and refers it to the other students to either support or refute. This would be coded 3, as the teacher is attending to the student's ideas and asking questions based on the student's ideas. Notice that a teacher question, based on student ideas is not coded 4.

Example 5) raises problems concerning synonyms, paraphrasing, abstracting, and subtle changes of meaning. The student did not say what the teacher said they did. The student's ideas were translated by the teacher into their own words, presumably in

order to push towards a certain desired response. The issue with this is whether or not the teacher is introducing a new concept, which would be coded 5, or whether they are building on the student's idea which would be coded 3. When a teacher makes use of an idea in such a way that it is probable the student would no longer recognise it as their own, it cannot be categorised as 3.

Category 4

Questions asked by a teacher which serve the purpose of driving the interaction along to a different step, to introduce a new problem element, and to include ideas which the teacher believes to be important, are coded 4. A second requirement is that the teacher acts as though they expect an answer.

Questions are usually fairly easy to recognise, even when they are open. Though not all questions are classified in Category 4, such as commands, praise or criticisms framed as questions. Teacher questions can really be coded in any one of the seven teacher-talk categories: in Category 1 if they are objective, nonthreatening enquiries involving attitudes or emotions and designate the feeling or emotion; in Category 2 if they are intended to praise; in 3 if they are based on ideas previously expressed by the student; in 5 if they are categorical and no answer is expected; in 6 if they are directions; and in 7 if they are intended to criticise, or to catch students who are daydreaming. Usually the questions which are coded 4 are genuine invitations to participate.

Category 5

Lecturing, giving facts, expressing opinions, interjecting thoughts, and off-hand remarks are all classified in this category. In a way, it is sort of a catch-all for teacher statements, primarily because it usually has the highest frequency, and an incorrect tally, more or less, would be least likely to distort the teacher's profile, compared to some of the other categories.

Categories 6 and 7

Both of these categories are used for statements which are intended to produce compliance. Category 6 is for statements which give direction to students and Category 7 is for statements which criticise the student. Such statements tend to enhance the authority of the teacher.

Examples:

- 6) "Please take out your books and turn to page 67."
- 7) "The trouble with you is that you don't follow instructions."
- 8) "The only work I'll accept is that which is neat."
- 9) "I want you to think about a holiday you've been on."
- 10) "Could you open the windows, please?"

Example 6) is a straightforward Category 6. 7) appears to be a straightforward 7, since the implication is that the student is being criticised for not doing something they ought. Example 8) could be either a 6 or a 7, depending on the utterance preceding it. If there was recrimination, it would be 7. Example 9) presents the problem of deciding whether the direction should be coded 6 when it is impossible to know if the command was followed.

Categories 6 and 7 are recorded to show close direction and supervision to the students by the teacher. Both categories help to establish a true teacher initiation- student response pattern. Even questions can be coded 6 as shown in Example 10) where the teacher issues a direction to the student to open some windows but frames this direction as a question.

Categories 8 and 9

Student talk is coded with these two categories. There are several dimensions which help to separate response from initiation.

One dimension which enters into separating 8's from 9's is the voluntary embellishment or enlargement of a topic. An aspect of identifying 8's and 9's is the contrast of indifference or conformity versus the expression of will through independent judgment.

Another factor in separating 8's from 9's is the element of creativity and higher mental processes compared with noncreative and lower mental processes. However, a student's statements can sound creative when they are merely repeated from memory. This is equally true of generalisations, and interpreting data.

Usually observers might like to infer, from the proportion of 9's to 8's in all student

talk, something about the freedom of students to express their own ideas, to suggest their own approach to a problem, and to develop their own explanations or theories.

Category 10

When there is a pause in classroom interaction, or when there is noise or confusion, Category 10 is used.

3.6 Evaluation of the Original FIACS (Relph, 2015)

In my previous work, I have examined the use of the original FIACS in ESOL classrooms. I conducted a study into the analysis of ESOL classroom communication using the original FIAC system, the recommendations from which formed the basis and motivation for this study:

That study found that the FIACS is an efficient tool with which to measure both the social and emotional climate of a classroom in a somewhat objective and reliable way. It can be used to provide feedback to pupils and teachers about classroom teaching. Because the FIACS focuses mostly on the teacher's talk, it is a good tool to use when comparing the different teaching styles of different teachers in the same subject, across, for example, different genders, nationalities and age groups.

Another key advantage is the that it is a very systematic way in which to analyse interaction, which allows it to be applied to large amounts of data in order to identify patterns of interaction. This framework can be used to identify the structures of classroom talk.

However, as previously stated, less attention is paid towards student-talk. As a result, interactions between the students are not considered here. Only teacher-student discourse is examined. There are many rules for observing classes correctly, as outlined in Flanders (1970), which must be followed in order to achieve accurate results with the FIACS. The system is also designed to be diagnostic in nature, as it can help to study the behavioural patterns of teachers and identify a teacher's 'strengths' and 'weaknesses'. It can also be used as a technique for the training of teachers. The analysis of the matrix is designed to be so dependable that even people who were not present at the classes are able to make accurate inferences about the verbal interaction. The entire process of using the system needs to be done by a trained observer, one who is familiar with the limitations of FIACS, for the results to be accurate. FIACS is a purely explanatory system, therefore it cannot be used to make judgements about what is good or bad teaching. The system is not designed to evaluate classroom interaction or teacher/teaching behaviour. FIACS does not describe the totality of classroom activity, although, as previously stated, no categorising system can ever be entirely unproblematic or comprehensive. So it is with full awareness of the problems and limitations of this system that I would like to propose some suggestions for a revised version to be deployed in an ESOL classroom interaction context. This is by no means intended to be an exhaustive list. My hope is that these recommendations are found to be of interest and could be build-able and/or trialled as a topic of future research.

3.7 Recommendations for Revisions to FIACS (Relph 2015)

My previous research into the use of the FIACS in ESOL-specific contexts has found that Flanders' system of one set of categories to be generalisable across all contexts is unlikely to work, given the context-dependent nature of all language. In this report, I am proposing a revised and updated category system to be deployed within an ESOL context. ESOL takes such a vastly different pedagogical approach to most high school methods of teaching, as well as encompassing a great deal of diversity within itself.

Perhaps where most of the previous attempts at formulating an interaction analysis category system, in the vein of FIACS, have suffered in the past, is their desire to create and devise one system to be applicable to all contexts. Indeed, Flanders (1970) has been, by far, the main player in teacher-student interaction analysis category systems. While other similar systems have been created, they have all sought to provide a definitive template of interaction analysis categories applicable in all situations. Perhaps this is not the solution. Indeed, it may be the case that we cannot have a generalisable system, because all language is very much contextually-based. Surely we must concede that it would be a mistake to attempt to map the conventions of one interactional context onto another interactional context and expect them to perfectly fit into each other and match up absolutely. An ESOL interaction analysis system would be complementary to Flanders, rather than competitive, as the two would operate in different spheres.

An ESOL interaction analysis system would have to address the issue of certain ESOL-

specific phenomena that FIACS does not currently accommodate, such as a way to categorise modelling and drilling.

Also, a way to distinguish between meaningful silences (i.e. those which exhibit misunderstanding and those which occur when the students are working quietly) would be useful.

One important aspect to achieve (perhaps the most important) would be that of making such a category system realistically employable. A balance must be struck between not constructing too many categories, making the learning of the system unnecessarily time-consuming, and not making the categories too broad (so that there is uncertainty on the part of the observer as to within which category they should label a particular interaction).

The adaption of the original FIACS involves a process of consultation with a group of ESOL teachers working in the UK. The researcher will meet with the teachers and interviewed them about the types of interactions that they experience in their classrooms and discussed the ways in which analysis of classroom interaction could be of benefit and interest to them as educators. Based on these consultations, the new category system will be drawn up taking provisions from the teacher's input and the existing literature. This preliminary version of the revised FIACS is used in a follow-up session with the same group of teachers who are given a sample transcript from which to test out the revised category system. After the teachers have attempted to use the category system themselves, they come back together with the researcher to discuss

what they liked and didn't like about the categories and whether or not they will have found the use of the test to be helpful or informative for them as teachers. The categories are then further adjusted based on this formative feedback.

3.8 Data Analysis Procedure

The data analysis for this study involved descriptive statistics using quantitative procedures. The classes were recorded and the recordings of each class were then transcribed orthographically, and coded according to the FIACS. The code reports of each class were then converted into a tally chart showing the frequency of each interaction in a pair. After all the data was collected and organised in the tally chart, it was then put into a matrix, and calculated into percentages in order to show the frequency of distribution between the different categories and to clearly provide an accurate picture of the interactional goings on of the classes.

In order to analyse the classroom interaction, data was recorded from the ESOL lessons by use of video cameras and microphones. After the lessons, the researcher played back the recordings and coded one symbol per every three seconds of interaction. After all of the data had been coded and categorised, each coded symbol was paired up with the symbol directly following it. All of the pairs of interactions were then counted up in a tally. Once all of the pairs of interaction have been tallied, the total frequencies of each interaction pair were combined into a table from which could be calculated the overall percentages of occurrence for each interaction type.

3.9 Conclusion

This chapter has described the methodology used in this study. Participants included 32 students and 14 teachers on an ESOL programme at a Northern English university. The study involved the collection of data from recordings of the classes. The procedures used to analyse the data were also described. The results of the study are presented in the next chapter.

CHAPTER 4: RESULTS (RESEARCH QUESTION ONE)

4.1 Introduction

The present study was designed with a twofold purpose: firstly to develop a system to categorise the interaction between teachers and students in an ESOL classroom context, and secondly, to investigate the interaction types most prevalent in a set of ESOL classes. This chapter begins with a rich description of the participants of the study and is followed by the results and discussion of the first research question:

1. Based on the recommendations outlined in Relph (2015), is the revised and updated FIACS an appropriate way to categorise ESOL classroom interaction between teachers and students?

4.2 Demographic Information

Course Participants

There were a total of 34 students participating in this study. One factor that may have impacted on the results of this study was the attendance of the participants in the ESOL classes, which greatly varied from class to class. One reason for this could be that the students were adults, who are older, and thus may tend to be busier, than traditional school-age students, and so were not able to attend all the classes.

Group A had a higher course attendance rate than Group B. There are several possible

reasons for this. Firstly, several participants in Group B mentioned that the scheduling of the course was not very effective, and that finding the time during the summer was not as easy as they had expected.

A second possible reason could be related to student motivation. While Group B signed up for the course at the same time as Group A, they started approximately three months later. The researcher speculates that this group was equally motivated when they signed up, but because there was a longer time lapse from when they signed up for the course to when they began, their motivation possibly decreased. Lambert (1991) and Diaz & Cartnal (2006) have suggested that shorter term length impacts course completion. Although these studies do not take into account the time involved in precourse preparation, the idea is the same: that a shorter time period creates an environment that helps to keep the students on task (Diaz & Cartnal, 2006: 3). A final reason why this course attracted such a high attrition rate in general was because the courses were offered for free and there was little incentive for students to keep attending the classes; no course credit was awarded and there was no loss of money involved in dropping out of the course.

Data was collected prior to the course, including age, gender, nationality and previous language learning experience. The information presented in the following tables describes the entire population of the study, including all course participants, because these demographic details will be important in the discussion of the results regarding factors that contribute to interaction in a classroom environment.

Participant Demographics

Table 4.1 shows demographic information for each student, including age, gender and nationality in Group A. Table 4.2 shows demographic information for each student, including age, gender and nationality in Group B.

Student	Gender	Age	L1
S1	Female	21	Chinese
S2	Female	34	Korean
S3	Female	19	Chinese
S4	Female	18	Chinese
S5	Female	25	Chinese
S6	Female	30	Chinese
S7	Female	23	Chinese
S8	Female	19	Chinese
S9	Female	21	Chinese
S10	Female	33	Spanish
S11	Female	19	Chinese
S12	Male	22	Chinese
S13	Male	27	Malaysian
S14	Male	20	Chinese
S35	Male	31	Brazilian
	Table 4.1- Gro	up A Demographics	
Student	Gender	Age	Nationality
S15	Female	28	Chinese
S16	Female	34	Chinese
S17	Male	33	Chinese
S18	Female	18	Chinese
S19	Female	22	Korean
S20	Female	24	Japanese
S21	Female	19	French
S22	Male	21	French
S23	Female	18	Japanese

S25	Female	23	Chinese
S26	Female	20	Chinese
S27	Male	24	Italian
S28	Female	21	French
S29	Male	23	French
S30	Male	30	Sri Lankan
S31	Female	33	Chinese
S34	Male	21	Italian

Table 4.2- Group B Demographics

The tables only show the demographic information for participants who gave consent to be included in this study. Participants who withdrew consent are not shown.

Student Gender

As shown in Tables 4.1 and 4.2, each of the groups had more female students than male students. There were 73% females in Group A and 65% in Group B. Overall, there were 69% females and 31% males.

Student Age

As shown in Tables 4.1 and 4.2, student age also varied. The majority of students (66%) were between the ages of 18 and 24.

Student Nationality

Students came largely from China. 56% of the students were Chinese.

Previous Language Learning Experience

Students also rated their language proficiency level. As discussed in Chapter Three, only participants who rated their English level as pre-intermediate to intermediate were accepted onto this course.

Teacher Demographic Information

As shown in Table 4.3, there was a greater number of female teachers than males teachers. Overall, there were 93% female teachers and 7% male teachers.

Teacher	Gender	Age	Nationality
T1	Female	20	English
T2	Female	20	English
T3	Female	20	English
T4	Female	20	English
T5	Female	20	English
T6	Female	20	English
Τ7	Female	20	English
Т8	Female	20	English
Т9	Female	20	English
T10	Male	20	English
T11	Female	20	English
T12	Female	20	English
T13	Female	20	English
T14	Female	20	English

Table 4.3- Teachers' Demographic Information

All of the teachers were native speakers of British English. They had studied for a degree in English Language and Linguistics from the same UK university. They had

also all obtained a qualification in CELTA no more than 12 months previously. The majority of the teachers had not taught a course since qualifying in the CELTA, but all had previous teaching experience from CELTA, and had taught using the BBC Speakout books on the CELTA course.

Summary: Demographic Information

There were a total of 49 participants in this study. 3 of these participants withdrew their consent leaving a remaining 46. Of the 46 remaining participants, 32 were students and 14 were teachers. 35 were females and 11 were males. Participants all resided in the place of study at the time this research was carried out, and had all had some previous experience of ESOL education prior to the study.

4.3 Research Question One

In this section, I present the results of the first research question, which asked:

Based on the recommendations outlined in Relph (2015), is the revised and updated FIACS an appropriate way to categorise ESOL classroom interaction between teachers and students?

This research question is a conceptual one. It examined the changes made to the FIACS in the revised and updated version and questioned whether or not these changes have improved the accuracy of the system in regards to ESOL-specific classroom contexts. What follows in this section are details of the revisions made to the FIACS and an evaluation of the appropriateness of its use in ESOL classrooms.

Revising the FIACS

The FIACS has previously been expanded for use in EFL by Moskowitz (1976) to contain over 20 new categories. However, it was decided for this study that the category system should be made more manageable by reducing the number of categories. By keeping a smaller list of distinct categories, this helps the system become easier to memorise and learn. The category system used in this study has been designed to be as practical as possible so that it can be used in an applied context. The methods described here can be used by teachers, teacher-trainers and education policy implementors, to similar effect, without the necessary need for a previous background in academic research. This hopefully allows the system to have cross-purpose appeal, in that it can be applied outside of an academic context. This increases the system's usefulness to people with an interest in classroom interaction.

The adaptations to the categorisation framework were devised in response to my own observations of the classroom interaction. By applying the original framework and then discovering there were elements of interaction that could not be adequately accounted for with the original, I amended the categorisations in order to make it more inclusive of those elements. The amendments to the categorisation system arose from my own critical engagement with, and application of, the framework to the data itself. This was followed up by a workshop session with the ESOL teachers who tested the revisions made and provided feedback on the category definitions.

The new category system consists of 17 categories, 11 which describe aspects of teacher-talk, 2 which describe aspects of student-talk and 2 which describe types of silence:

Category 1

Category 1 consists of teacher statements which accept and clarify an attitude or the feelings of a student in a non-threatening manner. The feelings may be positive or negative. This also includes predicting or recalling feelings. Flanders claims these statements are rare and infrequent. One reason for their low incidence is the use of a special rule which states that the teacher must literally name or otherwise identify the emotion(s) present before the statement can be classified as "1".

Category 2

Category 2 remains unaltered from the original version: it describes praise and encouragement by the teacher which carries the value judgement of approval. To look directly at a student and nod your head whilst saying, "Mm-mm"- with a certain inflection- is to communicate to the student that they are producing the correct behaviour and the teacher would like more of the same. The difference between Category 1 and 2, is that in 1 there is an element of objectiveness which is missing in 2. Both categories are used for statements which have overtones of warmth and friendliness, but Category 2 adds teacher approval as well. For the full outline of this category, revisit the description in Chapter 3. Category 3 also remains unaltered from the original version. It encapsulates several different teacher meanings:

- 1. They can acknowledge them through repeating them.
- 2. They can modify the idea and reshape it in their own words.
- 3. They can apply the idea and use it to infer the next stage of an analysis of a problem.
- 4. They can compare the idea to one expressed by another student or by the teacher.
- 5. Or they can summarise what was said by the student.

Category 3 occasionally raises problems concerning paraphrasing and abstracting which may cause a subtle change of meaning. When a teacher makes use of a student's idea in such a way that it is probable the student would no longer recognise it as their own, it cannot be categorised as a 3. This means that when teachers translate students' ideas into their own words, and at the same time add or subtract various meanings in order to perhaps push students toward a certain desired response, the idea no longer belongs to the student and becomes the teachers' own original idea. For examples of this, please revisit the full description of Category 3 in Chapter 3.

Flanders (1970) found that there is some evidence to show that greater incidence of Category 3 is associated with above average classroom measures of both student satisfaction and student achievement.

Category 4a

Questions asked by a teacher which serve the purpose of driving the interaction along to a different step, to introduce a new problem element, and to include ideas which the teacher believes to be important, are coded 4a. A second requirement is that the teacher acts as though they expect an answer.

Questions are usually fairly easy to recognise, even when they are open. Not all questions are classified in Category 4a, such as commands, praise or criticisms framed as questions. Teacher questions can really be coded in any one of the seven teacher-talk categories: in Category 1 if they are objective, nonthreatening enquiries involving attitudes or emotions and designate the feeling or emotion; in Category 2 if they are intended to praise; in 3 if they are based on ideas previously expressed by the student; in 5a if they are categorical and no answer is expected; in 6a if they are directions; and in 7a if they are intended to criticise, or to catch students who are daydreaming. Usually the questions which are coded 4a are genuine invitations to participate.

Category 4b

Elicitations made by the teacher which serve to prompt a specific response from a student, that the teacher views as desirable, are coded as 4b. As in 4a, the teacher must also act as though they expect an answer. The elicitations here refer to specifically "ESOL elicitations", a technique used by teachers to get the students to provide information rather than giving it to them.

Example 11)

The teacher elicits different types of food by asking students to look at some examples, then saying 'Some different types of food are...?'

In ESOL, elicitation is a vital component for a variety of reasons. It helps develop a student-centred dynamic, it makes learning memorable as students can link new and old ideas, and it can help produce a dynamic and stimulating learning environment.

Category 5

Lecturing, giving facts, expressing opinions, interjecting thoughts, and off-hand remarks are all classified in this category. In a way, it is sort of a catch-all for teacher statements, primarily because it usually has the highest frequency, and an incorrect tally, more or less, would be least likely to distort teachers' profiles, compared to some of the other categories.

Category 6a

Category 6a is for statements which give direction to students. Such statements tend to enhance the authority of the teacher.

Category 6a is used to show close direction and supervision to the students by the teacher. This category helps to establish a true teacher initiation- student response pattern. Even questions can be coded 6a.

Category 6b

This category is used for teacher directed drills or modelling and drilling, whereby the teacher models some language and students echo it back.

Modelling and drilling has been given its own subcategory because it is a feature which is much more common in ESOL contexts. Other non-ESOL classes may occasionally employ drilling, but not often the act of modelling and drilling together. At least not to the extent that it is prevalent in almost all ESOL contexts, and so is largely significant as a feature on its own. It has a fixed form and conventions which are noticeably distinct from the giving of other commands, and so deserves a separate category in the context of ESOL lessons.

Category 7a

Category 7a remains unaltered from the original version. It is for statements made by the teacher which criticise the student. Such statements tend to enhance the authority of the teacher. Category 7a is recorded to show close direction and supervision to the students by the teacher. It helps to establish a true teacher initiation- student response pattern. For the full outline of this category, revisit the description in Chapter 3.

Category 7b

Category 7b describes an utterance used by the teacher which seeks to mark their intention to create a boundary in the interaction. This occurs when the teacher wishes

to move the lesson along by changing topics or by starting a new activity.

Examples:

12) Teacher: So I want you to have a look at the text on the sheet.

13) Teacher: Okay, so, what answers have we got?

The function of "*So*" in both examples is to signal a change in the direction of the conversation, in this case to start a new part of the lesson, "so" is a discourse marker here. They can be used to focus, clarify, contrast, change the subject, show agreement or disagreement.

Category 8

This category describes talk by students which occurs in response to the teacher. Teachers initiate contact, or solicit the students' statements, or structures the situation. The students' freedom to express their own ideas is limited.

Category 9

Category 9 describes utterances made by students which demonstrate an expression of students' will through independent judgment. They contain an element of creativity and higher mental processes compared with the noncreative and lower mental processes of Category 8.

An observer might, from the proportion of 9's compared to all student talk, like to infer something about the freedom of students to express their own ideas, to suggest their own approach to a problem, and to develop their own explanations or theories.

Category 10

When there is a pause in classroom interaction, or when there is noise or confusion, Category 10 is used. There is very little point in recording a series of 10's for longer than 1 or 2 minutes. The FIACS is intended for situations where verbal exchanges exist between teachers and students, or in which such an exchange is imminent.

Category 11

Another shortcoming of FIACS is that transactional silence is not distinguished from silent working. An 11th category has been added and used to make distinctions between silences which are of interest to the teacher.

This category is for silence where interaction is no longer expected. This could be because the students are silently working and so are not engaged in interaction at that moment, and neither is an interaction imminent. This is separate from Category 10 which has been revised to only include transactional silences, that is, silences where further interaction is expected such as short pauses in speech where the interactant has not finished their turn. Under the new revised system, when an interchange no longer is expected the observer switches to the new Category 11, as opposed to the original study where they simply stopped categorising. It may be helpful to remind ourselves at this stage that the FIACS was not designed to answer detailed questions about different types of silences. By developing only two out of ten categories for student-talk, and one to silence, the inferences about student-talk and silence that can be drawn will be limited.

Table 4.4 The categories of the modified FIACS with the new revised editions of the categories highlighted in bold (Adapted from Flanders, 1970).

	Response					
	1. Accepts feelings. Accepts and clarifies an attitude or					
	the feeling tone of a student in a nonthreatening manner.					
	Feelings may be positive or negative. Predicting and					
	recalling feelings are included.					
	2. Praises or encourages. Praises or encourages student					
	action or behaviour. Jokes that release tension, but not at					
	the expense of another individual; nodding head or					
Teacher Talk	saying "Um hm?" or "go on" are included.					
	3. Accepts or uses ideas of students. Clarifying, building,					
	or developing ideas suggested by a student. Teacher					
	extensions of student ideas are included but as the					
	teacher brings more of their own ideas into play, shift to					
	category five.					
	4a. Asks questions. Asking a question about content					
	or procedure, based on teacher ideas, with the intent					
	that a student will answer.					
	4b. <i>Elicitations</i> . Teacher prompts a specific response					

	from the student.						
	Initiation						
	5. <i>Lecturing</i> . Giving facts or opinions about content or						
	procedures; expressing <i>their own</i> ideas, giving <i>their own</i>						
	explanation, or citing an authority other than a student.						
	6a. Giving directions. Directions, commands, or						
	orders to which a student is expected to comply.						
	6b. Modelling and Drilling. The teacher models some						
	language for the students to echo back.						
	7a. Criticising or justifying authority. Statements						
	intended to change student behaviour from an						
	unacceptable to an acceptable pattern; bawling someone						
	out; stating why the teacher is doing what they are						
	doing; extreme self-reference.						
	7b. Marking discourse.						
	Utterances by the teacher which are used to mark						
	discourse. They are used to focus, clarify, contrast,						
	change the subject, show agreement or disagreement,						
	etc.						
Student Talk	8. Student-talk—response. Talk by students in response						
	to teacher. Teacher initiates the contact or solicits student						
	statement or structures the situation. Freedom to express						
	own ideas is limited.						
	9. <i>Student-talk—initiation</i> . Talk by students which they						
	initiate. Expressing own ideas; initiating of a new topic;						

	freedom to develop opinions and a line of thought, like						
	asking thoughtful questions; going beyond the existing						
	structure.						
Silence 10. <i>Interactional Silence</i> . Silence that is parts of an interactional sequence. Pauses,							
10. Incluctional Sucheet Shenee that is parts of an interactional sequence. I auses,							
short periods of silence and	periods of confusion in which communication cannot						
short periods of silence and be understood by the obser							
be understood by the obser	-						

Note that there is no value attached to these numbers. Each number is purely classificatory, to designate a particular communication event, not to pass a judgment on that communication.

Evaluation of the Revised Category System

interaction with the teacher is imminent.

Since classroom interaction is so complex and involves many nuances, category definitions and ground rules, the definitions of each category can never completely cover all the problems that will arise. However, there are many differences in pedagogy between ESOL teaching and ordinary native language speaking practices, that the FIACS does not readily account for. This can lead to seemingly unusual results when applying FIACS to an ESOL context and issues about administrating it in the first place. Flanders gives permission for one to "feel free when you investigate patterns of classroom communication, to develop a different procedure which matches your problem or modify the procedures [described]" (1970: 77). This idea will be expanded upon in this next section.

The revised interaction analysis category system is designed to be an efficient tool with which to measure both the social and emotional climate of an ESOL classroom. It provides a method of examining and recording interaction between teachers and their students in a way that is objective and reliable. This revised interaction analysis category system could be used by ESOL teacher trainers as a way in which to provide feedback to trainee teachers about classroom teaching, as well as by teachers to provide feedback to their colleagues, or for self evaluation. Because the revised interaction analysis category system focuses mostly on the teacher's talk, it is a good tool to use when comparing the different teaching styles of different teachers in the same subject, across, for example, different grade levels, genders, nationalities and age groups.

However, as previously stated, less attention is paid towards student-talk. As a result of this, the interactions that occur between student and student are not considered here. Only discourse involving both the teacher and students is examined. There are rules for observing classes correctly using this system, which must be followed in order to achieve accurate results with the revised interaction analysis category system. The revised system is also designed to be diagnostic in nature, as it can help provide insight into the behavioural patterns of teachers, to identify and study a teacher's relative 'strengths' and 'weaknesses'. It can also be used as a way to measure and evaluate teachers in training. The analysis of the matrix is designed to be so dependable that even people who were not present at the classes are able to make accurate inferences about the verbal interaction.

The entire process of using an interaction analysis category system needs to be done by a trained observer, one who is familiar with the limitations of such a research method, in order for the results to be accurate. This revised interaction analysis category system is intended to be taken as a purely explanatory device; therefore it cannot be used as a means of distinguishing good and bad teaching behaviours. This system is not designed to evaluate classroom interaction or teacher/ teaching behaviour. This revised interaction analysis category system does not describe the totality of classroom activity, although, as previously stated, no categorising system can ever be entirely unproblematic. So it is with full awareness of the problems and limitations of this system that I would like to explore, in greater depth, the alterations made to the categories and their definitions in order that this system is now better equipped to be employed in an ESOL specific classroom context. This will hopefully provide some justification and insight into the reasoning behind why the FIACS was adapted in this way. My hope is that these adaptations are found to be of interest and could be buildable and/or trialled as a topic of future research.

Advantages and Disadvantages of using Interaction Analysis Coding Systems in the ESOL Classroom.

There are advantages and disadvantages to using Interaction Analysis Coding Systems such as this in an ESOL classroom. One of the advantages, indeed perhaps the main advantage of using such a system as this is that, in simple applications, coding requires only a pen and paper. This revision of the FIACS sought to keep the categories distinct and all-inclusive, without compromising on usability. What is meant by this is that this system has been designed to be used by people such as ESOL teacher or teachertrainers who do not necessarily have a background in classroom interaction or conversation analysis; indeed even in linguistics. The categories themselves have been designed so as to be easy to understand and easy to memorise with no specialist technical skills training required beyond a need to learn the category definitions. This means that this system has a practical application and there are opportunities for teachers to employ this interaction analysis category system within their own classrooms.

The display of results can be designed to compare the results of the coding process with other data, or with a coded model. Observation is systematic, programmed, and its reliability can be determined. Long sessions of classroom teaching can be efficiently displayed.

On the other hand, accomplishing any sort of skill in observation requires practice. Although effort was taken in the designing of this revised system to make the categories distinct and therefore straightforward to learn, it would still take some effort on part of the observer to familiarise themselves with each of the category definitions in order to sufficiently learn the categories and to feel confident about making professionally important discriminations. In some instances, tedious clerical work is required when it comes to getting the figures ready to be analysed. This revised interaction analysis category system is designed to be applicable in ESOL classrooms and was only tested in ESOL classroom at the development stage. Usage in other FL classroom settings may be possible but are untested, and could provide a topic of further research. This interaction analysis category system is not intended for use in non-FL settings as this may have a negative impact on the actual interactional significance of the results. For non-FL classroom contexts, the original FIACS is still best applicable.

4.4 Conclusion

This chapter has outlined the demographic information of the students and teachers who took part in this study. It also explains the revisions made to the FIACS and how these revisions will strive to ensure that the FIACS can have a greater accuracy in portraying the interaction taking place in ESOL classrooms.

Research Question One sought to examine the changes made to the FIACS in the revised and updated version and question whether or not these changes have improved the accuracy of the system in regards to ESOL-specific classroom contexts.

ESOL teaching takes such a vastly different pedagogical approach to most non-ESOL methods of teaching, as well as encompassing a great deal of diversity within itself that having a separate category system to categorise interaction within this context seemed necessary in order to provide a truly accurate reflection of interaction that occurred within this context, as there are certain ESOL-specific features of interaction that were not adequately covered under the old FIACS. This revised version of Flanders' interaction analysis category system is intended to be a specialised system for use in categorising interactions in the ESOL classroom. This study assumes the contextually-based nature of language and so, based upon that assumption, a context generalisable interaction analysis category system cannot work accurately.

The revised interaction analysis category system addresses the issue of certain ESOLspecific phenomena that the FIACS does not currently accommodate, such as modelling and drilling.

The FIACS Category 10, which had previously encompassed all silence and confusion in the classroom, has now been split into two categories, 10 and 11, for interactional silences and transactional silences respectively.

CHAPTER 5: RESULTS (RESEARCH QUESTION TWO)

5.1 Introduction

In this chapter, I present the results of the second research question, which was designed to explore what an interaction analysis category system can reveal about the types of interaction which occur between teachers and students in ESOL classroom contexts.

5.2 Research Question Two

What does an analysis of student-teacher interaction, using the modified Flanders Interaction Analysis Category System, reveal about the different interaction types produced by the teachers and students in the context of an ESOL classroom?

The second research question sought to provide a picture of how students and teachers interact with each other in an ESOL class as part of their language learning. The results for Research Question Two include the entire course population; all research participants are included because the different behaviours of participants can shed light on the overall interactional "picture" of a class. An overview of the results of all the research questions can be found in Chapter 7.

Very little is known about the actual practices of students and teachers in an ESOL education environment. A majority of the literature written is speculative, hypothesised or based on student-teacher perception of non-ESOL classrooms rather than on actual ESOL practices. Research Question Two therefore offers a unique, detailed insight into the interactional practices of ESOL students and teachers. In addition, much of the research available on interaction is by and large inferential, and is not descriptive in nature.

Interactional Purpose

Each utterance on the course was coded as having one of 14 interactional purposes. The majority of utterances made were in direct relation to language learning tasks.

In a study of learner preferences, Hao (2004) found that students preferred instructional interaction (Revised Category 6a/ FIACS Category 6) more than any other types of interaction; she defined this as anything considered to be directly related to the course content. Lie (2008) correlated this to the combination of foreign language and language and cultural tasks, which students did complete more than any other tasks. So although Hao's study was based on students' attitudes towards the task, the same results were found investigating actual practices: that students prefer to spend time on the course content.

In summary, participants interacted with each other in different ways, although some distinct patterns arose.

5.3 Observing Classrooms Using the Revised FIACS

The following section will examine the data collected in Research Question Two in greater detail in order to better understand the interactional practices that occur

between teachers and students in ESOL classrooms.

Although everything that happens in a classroom depends on the coordination and cooperation of the teacher and students, it is usually considered the norm for the teacher to be in control. The teacher ordinarily makes most of the managerial decisions, and decides who is allowed to talk, to whom, what they will talk about, in what language and so on.

 Table 5.1 Frequency of Interaction Categories

Cat	1	2	3	4a	4b	5	6a	6b	7a	7b	8	9	10	11	Tot
															al
Tot	44	440	975	1639	163	3973	830	60	26	605	2593	3154	1032	571	1610
al															5
%	0.2	2.73	6.05	10.17	1.0	24.66	5.15	0.37	0.16	3.75	16.1	19.58	6.4	3.6	100
	7				1										

Category 1

0.27% of classroom interaction is spent on understanding and accepting students' feelings. Examples of teacher talk observed in this study that fall into this category would be:

Teacher 5- "Are you okay Student 3?"

Teacher 1- "You're welcome hm."

2.73% of classroom time is spent on encouragement of students. "Feedback is an inevitable part of classroom interaction" (Chaudron, 1988:133). Previous research has shown that positive feedback is more helpful in improving students' learning than negative feedback (Nunan, 1991). The same research also showed that simple praise, for example, "good", or "very good", does have a positive effect on students' performance. In the classes observed for this study, praising students and/or accepting their answers was a common feature of almost every lesson. Neglecting was not observed in this study. Cardelle & Corno (1981: 260) studied instructor feedback in a FL course and found that feedback leads to the development of linguistic forms, stating that it:

...supports[s] the notion that a major learning function can be served, at least in second language acquisition, by feedback that makes students' errors salient in a motivationally salient way. Specific feedback on errors draws attention to material not adequately learnt, allowing the student to focus there and not be distracted by too much re-examination of work done well.

The students in this study could find feedback particularly helpful, especially because students in this environment are likely to be much more extrinsically motivated to master the language. They are taking the language to complete a language requirement- they live in an English speaking country and many of them are here to work or to study. In addition, 6.05% of classroom interaction is spent by the teacher, on acceptance of, or benefiting from students' comments and elaboration of their opinions.

The following is an example of a Category 3 teacher-talk followed by a Category 2:

Teacher 8- "Wearing waterproof clothes, well done."

Student 7- "Suitcase."

Teacher 8- "Suitcase, well done. Anything else?"

The teacher accepts the student's utterance of suitcase by repeating the utterance showcasing an acceptance of the student's idea (Category 3), then follows by immediately praising the student with a "well done." (Category 2) This combination of 3 followed by 2 was a common co-occurrence of categories observed in this study.

Category 4a

10.17% of time belongs to the questions asked by teachers.

Teacher 5- "Yeah, which picture is aspirin?"

Students- "O."

The teacher asks a question to which the students respond with an answer. Such an utterance provides the students with an opportunity to take an active part in the interaction, albeit in a structured, controlled way that does not allow for the forming of talk independent from the teacher.

Category 4b

Another way that students have an opportunity to take an active part in the interaction, in a structured way is through answering teachers' elicitations. Elicitations in this study are classified according to what we define as "ESOL Elicitations" where the teacher gets the students to provide the information rather than giving it to them. Examples of this can be seen in the following interaction:

Teacher 8- "Binoculars."

Students- "H."

Teacher 8- "Dictionary."

Students- "K."

Teacher 8- "Digital camera."

Students- "C."

Here the teacher is eliciting the answer from the students to the question of which picture represents which object she is saying. Rather than simply saying 'The binoculars are in picture H', the teacher states the names of the objects and elicits the response from the students. This allows students opportunity to interact with the teacher in the task.

Category 5

The results obtained from this study indicate that of all interactions occurring during ESOL classroom processes and lessons, 24.66% belong to teachers' speeches and explanations. Even though Category 5 still accounts for the largest proportion of class-talk-time, the difference between that and Category 9 is not as large as the previous studies would show. Flanders (1970: 52) states that "normally 4, 5 and 8 are the most frequently occurring categories." Thus, we can expect there to be more 5's, 6's, 7's and 8's over 1's, 2's, 3's and 9's.

Teacher 6- "Okay, so to travel light means to not take any items, so take everything in one bag, one small bag, so if you're travelling light you're not taking a lot with you, okay?"

Teacher 10- "Okay, well we would like to go to Easter Island. It is very isolated, very far away from other places and the nearest country is Chile. Over two thousand miles away."

Teacher-centred methods encourage students to digest vocabulary and grammar, but this can limit their ability to apply their knowledge and formulate their own unique responses. Cole and Chan (1994) believe that more student participation plays a fundamental role in a successful lesson. Using a whole-class discussion style of learning means that every student is kept in a state of absolute consciousness during the lesson, so they are likely to remember more of the lesson content because they played an active part in it. If a teacher tries to deliberately avoid class discussions to keep student-talk to a minimum, this means that students are more likely to day-dream and not take as much of the lesson in. A teacher needs to direct the interaction to perhaps a specifically targeted student in order to keep learners in the state of absolute consciousness during class, such as randomly nominating students to speak. In order to increase student-talk in the classroom, the teacher can help to create more opportunities for student interaction. For example, the teacher could incorporate more problem-solving tasks into the lessons. Activities such as information gap exercises and those which involve the students collaborating together would foster more interaction. These tasks would all encourage greater oral output, number of turns taken in spoken language, and negotiation of meaning.

Category 6a

The third largest ranking of classroom interaction belongs to instructions and behavioural questions asked by teachers of students, i.e. 10.17%.

Teacher 3- "Can you just read through that on your own for now and answer these two question if you can do?"

Teacher 7- "So listen carefully and see if you can hear the different sentences in the box at the bottom."

Category 6b

Modelling of target language by the teacher is a specific feature of ESOL lessons. Here, in this study, it made up 0.37% of interaction time in class.

Teacher 10- "The trip is going to take."

Student 12- "The trip is going to take."

Teacher 10- "The trip is going to take. Just repeat--"

Student 14- "The trip is going to take."

Teacher 10- "Some of the problems we're going to face include."

Students- "Some of the problems we're going to face include."

In these examples observed, the teacher is modelling the desired language and the students repeat it together after the teacher. This technique is known as modelling and drilling and is a technique taught on the CELTA course, of which all of these teachers trained, to help students learn pronunciation.

Category 7a

Teachers spend 0.16% of interactions on severe criticism of students and imperative justifications.

Teacher 2- "Okay everyone, just the last thing before I go, Guys!"

Student 2- "Sorry"

Teacher 2- "Can everybody look here? I know Student 2 wants to have a chat but she should wait! Rude!"

Criticism of students' behaviour was rarely observed in this study, possibly due to the fact that the students were all adults and so were able to behave themselves well in an adult manner, cooperating with the teacher and their fellow students. Most of the time the students were actually older than the teachers, so it is possible that on the occasions where teacher wished to change the behaviour of the students, they felt unable to, out of embarrassment or fear of being disrespectful. Teacher-talk time may have also been affected by the fact that the teachers were newly qualified and did not have considerable teaching experience. For example, giving oral feedback and error-correction are areas that ESOL tutors gain confidence in as they become more experienced in the field.

Category 7b

Category 7b describes an utterance used by the teacher which seeks to mark their intention to create a boundary in the interaction. This occurs when the teacher wishes to move the lesson along by changing topics or by starting a new activity. This made up 3.75% of classroom interaction in this study.

Teacher 5- "Okay, so, how about we read the extract?"

Teacher 9- "So, what does the first text say?"

The function of these utterances is largely to signal a change in the direction of the conversation, in both of these cases the teacher is attempting to focus the students onto a task or topic.

Category 8

16.1% of interaction belongs to students' talks pertaining to teachers' questions. Kearsley (1999) stated that although it is a widely held belief that a high level of interaction is desirable and positively affects that effectiveness of education, it is not clear from the research or evaluative data that interaction improves that quality of instruction in most education programs. Jiang & Ting (2000) have specifically suggested that receptive interaction is an important source of learning and that the completion of tasks is much more important than once thought. Bandura (1986) differentiates learning from performance and argues that individuals can learn through observation alone, which could lead to learning without even showing performance. Beaudoin (2002) found that "lurkers" still feel that they are learning and that they are benefiting despite keeping a low-profile approach in the classroom, and that students who are highly interactive do not necessarily achieve better grades. However, the research in this area is highly speculative and not based on actual student practices.

Teacher 2- "Yeah. So when it was raining the girls decided to escape?"

Student 10- "Yeah, the girls decided to escape."

Teacher 2- "Can you find the past simple past continuous sentence in that?" *Student 2-* "Past er simple? Before he became a famous rev er revolutionary" In this example all of the students' utterances are made in direct response to the utterances made by the teacher. The students do not attempt to change the subject or tract of the interaction, which is driven by the teacher. Carroll (2001) maintains that it is still not certain what exactly interaction provides, and if it is anything more than practice. These results could also reflect Meunier's (1998) finding, that student anxiety is initially very high in ESOL classes. Students who are just getting accustomed to the nature of FL learning might need more time to develop a comfort level with the classes. They must first familiarise themselves with arriving in a new country, familiarising themselves with a new language and a new culture, and then with the collaborative, communicative-centred tasks that may not have been reflective of their previous FL learning experience in their own countries. This course may not have been long enough for some students to feel 'safe' completing tasks that do not resemble their previous language learning experiences. Also, the students were taught by many different teachers and so did not get the chance to cultivate a relationship with the sane teacher over multiple lessons, which may have meant that building a rapport in the classroom was harder.

Flanders' original study examined interaction processes between teachers and students between the ages of 12 and 17. Because little of the research on FL learning focuses on adult learners, students in this context may be mature enough to be able to examine the language systems, and are potentially more focused on mastering the grammatical structures of the language, possibly reflecting how they learnt language previously. The majority of the students observed in this study came from a Chinese background where cultural differences with regards to interaction preferences and styles may have a significant effect on preferred learning styles. For example, Jin and Cortazzi (1998) and McKay et al. (2008) looked at Chinese ESOL students with British teachers and found that the Chinese students did not see classroom discussion and interaction as being of much importance to their language learning. Instead they placed a heavier emphasis on teacher-directed lectures.

These findings also mirror that of Curtis et al. (1999: 44) who suggest that "some language students, particularly those who are accustomed to teacher-centred approaches may believe that they cannot learn from their peers or from a more collaborative style of learning." Thurmond et al. (2002), have also reported that students report less satisfaction working with their classmates.

These results could also take into account Knowles's (1998) distinction of andragogical and pedagogical learning principles (e.g., how adult learners learn differently from children). Knowles makes several distinctions between the needs of adult learners from the needs of younger learners. First, is the distinction of understanding why something is important. For FL learning, this could transfer into learners needing to understand why the language is structured the way it is, and why they need to use a specific language structure in a specific sentence type. Another point Knowles makes is that learners need to relate the materials to their previous experiences. Because the students are all living in the UK, discussing UK culture and society, as well as the topic of travelling, could be topics of specific interest to them.

The input hypothesis suggests that the completion of receptive tasks should lead to linguistic development (Krashen, 1985) and the output hypothesis suggests that

productive tasks would lead to linguistic development. While these results seem to indicate that interaction is a vital part of language learning, other findings have not shown that a relationship between interactivity and learning outcomes exist (Ebner & Holizinger, 2005). Freeman & Anderson (2007) studied "lurkers" and found that active participation in class discussions had no relationship to higher grades.

Additionally, Kearsley (1995) commented that if a student was strongly autonomous and extremely self-driven, they might not need as much interaction than other less independent students. However, these studies all examined interaction in classrooms quite generally, and were not language learning focused. The Sociocultural theory, especially, at its core, posits that learning cannot be done in isolation; learning has to be done within a social context. The Interactionist theories posit that language learning is a social process, which "evolves out of learning how to carry on conversations" (de Bot, Lowie & Verspoor, 2005: 404).

Category 9

19.58% of time belongs to students' free talks and the talks that are initiated by students. This shows that students in the ESOL classrooms observed for this study talk much more independently of their teacher, and take the initiative more than in previous research examining non-ESOL classrooms (Walker, 2002; Hai & Bee, 2006). In other words, the ESOL students in the classrooms observed in the course of this study are active in the classroom discourse. Furthermore, their participation is more aligned with communicating in a more creative, initiative-taking and freer type of talk, rather than just talking in response to the teachers' utterances.

Student 2- "Excuse me, travelling, is it a double L or..."

Teacher 2- "Is travelling double L?"

Students- "Yes."

Student 2- "You don't like activities?"

Teacher 3- "Not in the snow, I don't!"

It might have been expected that the proportion on student-talk would be higher than in previous studies because of the way in which ESOL lessons are designed to deliberately give students more opportunities to speak and practice their language. In a sense, I think that is what is shown here. In this study, student talking time, spent in free and initiative-taking talks is noticeably higher than the results obtained in the study by Sahlberg (2008).

The greater prevalence of Category 9, in the data collected for this study, over Category 8 is interesting and unusual, demonstrating that teaching practices in the ESOL classrooms observed here tend to be more student-centred than normal classroom contexts and ESOL students take a much more active role in classroom interaction.

Active participation of students is greatly encouraged in the ESOL classroom, with the majority of lessons designed specifically to give students opportunity to practice their speaking and listening skills. Ahmad & Aziz, (2009) and Odundo (2013) show that by

making lessons more student centred and having less teacher-talk time, the students can actually learn more. They suggest that, student-centred methods may be more effective in enhancing learning achievement than teacher-centred approaches, but only in certain educational contexts. Student-centred methods are known for stimulating innovation, critical thinking skills and retention of knowledge, whereas teacher-centred methods carry the association of limited opportunity of innovation and a more scientific-based approach to thought.

Another explanation for the prevalence of student-centred talk could be that the participants of this study were all students who genuinely wanted to learn. There were no guidelines or requirements for participating in lessons, students could attend (or not attend) as many or as few classes as they wished. However, as the classes ran were offered for free; and there being no other similar classes offered free in the area, it is possible that this could have made some students feel as though they did they not have any other options available for language learning and so made as much of an effort to participate in class as possible.

A study by Jiang & Ting (2000) found that teacher expectations for interaction were linked to perceived learning. In this study however, there were no guidelines or requirements for participation; learners participated in as many or as few lessons as they found beneficial to their learning.

Another explanation for this finding is that participants who genuinely wanted to learn English had little other options similar to this one available in the area at the time. In the location of the study, there are English Language courses available, but even though they may be in the same city, there are possibly some conflicts due to scheduling, fees, etc. It is likely that this could make some participants feel a certain degree of urgency; those who really felt that they did not have any other options for language learning other than these free classes likely made an effort to do as much as they could during the duration of the course.

One of the weaknesses of the interaction analysis category system used in this study, it has to be said, is that 9 is the only code symbol which can be used for off-hand remarks by students, counter-dependant statements, and resistance to compliance. That is, both cooperative and uncooperative initiation falls into the same category. This may have consequences for the interpretation of data such as the perception of students challenging teachers' authority in the classroom, and student satisfaction levels which researchers may be able to draw preliminary inference from distinguishing between initiation that is cooperative or uncooperative. Also examining students' negative or positive behaviour is only possible by examining the reaction of the teacher to that behaviour, not by being able to see the behaviour itself.

Category 10

10% of class time is spent in interactional silence. Silence or confusion, where the observer could not see any interaction between teachers and students, was recorded. Previous research on silence in the classroom has found that the percentage of time in the lesson where silence occurred was around 1% (Atakin & Brown, 2001; Anorue, 2004; Schulz, 2010). Having 1% of the total class time observed be silence could seem

to support the view that most teachers dominate classroom discourse, if it is in the fact the teachers who are doing most of the talking.

The FIACS Category 10, which had previously encompassed all silence and confusion in the classroom, has now been split into two categories, 10 and 11, for interactional silences and transactional silences respectively. It was felt that there should be a distinction between silences which occur as a result of a pause in interaction, where interaction is expected (interactional silences), and silences which occur as a result of students perhaps completing an activity in silence. Therefore the silence occurs not as a result of nothing happening in the classroom, but just as a result of interaction between teachers and students not being required for a certain specific activity (transactional silences).

One factor to do with the prevalence of interactional silence in this study, may have to do with the level of this course. Most research on classroom interaction involves learners at advanced levels of FL study. Students might not feel as confident or may be deficient in their language level to interact with the teacher at an intermediate level. Students in most FL classes have time to establish a rapport with one another and their teacher, whereas in this study they had different teachers every week and different fellow students.

Category 11

There was a high proportion of classroom time taken up by transactional silence on behalf of the participants. There are several explanations for this result: Firstly, the nature of some of the tasks focused on discrete language skills rather than communicative competence, and so the interaction level would be lower during those tasks. In the case of this study, the silence in the classroom occurred very often in the context of students being instructed to complete an activity in silence. There may have been an absence of verbal interaction between the teachers and students, but the vast majority of the time this was caused by the students quietly completing their work with the teachers monitoring close-by. In all lessons observed, the teachers were ready to interact by answering questions, if needs be.

Schulz (2010) argues that silence occurs on a continuum; and to understand the role that a particular silence plays for the individual interactant, as well as the class as a whole, is a nuanced and complex process that may require new ways of conceptualising what is meant by listening. In the past, and certainly when Flanders was conducting his initial observations, silence was categorised as the absence of interaction. However, silences in the classroom, indeed silences everywhere, are capable of carrying different meanings dependant on their context and in fact can be a vital part of interaction. Schulz (2010) further emphasises that teachers should reconsider their silences, by carefully listening to their students and enquiring of them, which in essence, will shift teachers' understanding of students' participation. Schulz therefore concludes that participation should be redefined to include silence, a conclusion with which this thesis concurs.

Additionally, some learners may feel "safer" completing individual tasks rather group tasks. This security for individual tasks might be for one of several reasons. First, individual tasks are private and no one knows when the student has made an error (except the student), so they do not need to build up a comfort level with each other prior to completing them. Also, as the majority of students came from a Chinese background where greater emphasis is placed upon teacher-centred lecture based lessons rather than student-centred discussion based ones, students may feel that more discrete language learning items are "safer" because it is the type of learning that is familiar to them; it more closely resembles the language courses they have previously done. So interacting lots with other students may feel slightly awkward or unnatural.

In other words, 34.09% of interactions pertain to teachers' direct influence and 20.23% to teachers' indirect influence, i.e. 54.32% of total classroom interactions pertain to teacher talk. Based on these results, one can surmise that teachers are wont to use direct influence on students (Gross, 1993; Weimer, 1993; Anorue, 2004; Orunu, 2012). In addition, 35.68% of total classroom interaction pertains to student talk.

While many previous studies (Amidon & Hunter, 1967; Flanders, 1970; Walker, 2002; Wragg, 2002; Fathi- Azar, 2003; Hai & Bee, 2006; Blatt *et al.*, 2008; Hafiz-Mahmud *et al.*, 2008) concluded that teachers' direct talk time was around 75% of their total talk time in non-ESOL classrooms, the results obtained in the present study show a lower ratio. This indicates that students' participation in classes, in an ESOL context, is higher than in similar studies conducted outside of an ESOL context.

As noted in Flanders (1970), the teaching approaches favoured by teachers are influenced by the objectives of the lesson and by teachers' own individual teaching styles. The results of this investigation show that teachers did not stick to one specific method of delivery, showing great variation in styles. Most of the teachers used a variety of instructional methods, whether this was done consciously or unconsciously.

5.4 Conclusion

This chapter has presented the results from descriptive data (interactional practices) and from the inferential data (the effect demographics may have on interactional practices). An overview of the interactional practices are presented in Table 5.1.

Research Question Two sought to provide a picture of how ESOL teachers and students interact in an ESOL course. The results for Research Question Two include the entire course population; all course participants are included, not solely the course completers, because the different behaviours of participants can shed light on the possible benefits. or not, of interaction.

The results suggest that if learners are provided with a rich learning environment, including a variety of tasks they can freely choose from, they will complete tasks that they consider to the most beneficial to their language development. In the present study, participants completed an average of 96 tasks during the eight week course. More tasks were completed during week one than any other week; the fewest number of tasks were completed in week 8. there were a comparable number of written and oral activities completed. Another significant finding is that that the interactional practices of the groups varied slightly. Participants were divided into two different groups for the study. These groups were not given different treatments, but the second group (Group B) seemed to have a group dynamic that was very strong and several of the participants appeared to know each other from before the course had started. This helped to get conversations started, and over time, generated a strong sense of community. This was not observed to the same effect in Group A. Group A also had a higher rate of students not turning up classes than Group B. One possible reason that Group A did not have a higher student retention rate might have been because of the timing of the course, as the Group A course was run quite close the Christmas period.

Some of the findings from these analyses resemble previous research conducted in non-ESOL studies, but some are unique to this study possibly due the ESOL context, the community program in which students are enrolled, or that previous research is based on perceptions of interactions (Hao, 2004; Kearsley, 1999), not on actual practices (Freeman & Anderson, 2007; Paulus, 2007).

CHAPTER 6: RESULTS (RESEARCH QUESTION THREE)

6.1 Introduction

In this chapter, I present some thoughts concerning the third research question, which was designed to explore the ways in which an interaction analysis category system could be practically applied in the training and development of ESOL teachers.

Research Question Three

How could an interaction analysis category system be practically applied in the training of ESOL teachers?

6.2 Practical Applications of Interaction Analysis Coding Systems in Teacher-Training

One of the main aims of this study was to produce a framework which could be employed by teachers in ESOL contexts in order to monitor their levels of interaction in the classroom. In order to make this system realistically and practically employable, it should be designed in a way that makes it accessible to those teachers who do not necessarily have an academic background in classroom interaction, conversation analysis, or even linguistics. In order to ensure this, the revised framework was tested by teachers with no previous acquaintance to the FIACS in order to gain their feedback into how user-friendly the new system was, how easy was it to understand and implement, and what criticisms or changes did they feel could improve the system. Originally, it was felt by the teachers that having an extra category to show discourse markers was important in order to distinguish them from lecturing, and so this category as added.

The teachers felt that overall the system was straightforward and clear, with good distinctions make between each of the categories so that none of them overlapped one another. The descriptions of some of the categories were also made more explicit based on this consultation. The teachers each separately categorised a sample transcript from one of the ESOL classes observed in this study, and when comparing their classifications alongside that of the researcher, reached a categorisation consensus of around 70% agreement with the researcher. This demonstrates very clear proof that this revised category system is able to be used to a reasonable level of success by teachers with no previous experience of the FIACS.

Teachers are always looking for ways to improve in the classroom. The data collected on practitioners' engagement with the categorisation system caused the practitioners to reflect on how they could increase students' English Language production and interaction in lessons. The teachers found this useful to be able to see when and where the student interaction was happening, and this led to them being able to set goals for classroom interaction by planning more interaction-friendly activities for their lessons.

> "There are two main things that help students learn English, [and that is] time and practice. Although you can't rush learning, you can provide [the students] with practice time. By using the revised

FIACS, I can see where I can improve activities to give more space for interaction. Hearing my students speak gives me the opportunity to gauge what they have learned, and it is a way I can monitor their progress."

-Comment from interview with 'Teacher 3'.

A potential side-effect of using the revised FIACS, is that by recording lessons for categorisation, teachers can use the information gathered to assess students use of target language because they will have a record of the talk occurring during a lesson. If students are having difficulty with phrases or vocabulary, the teacher will be able to offer guidance or further instruction to support language development.

Wong Fillmore & Snow (2000) state that:

Teachers play a critical role in supporting language development. Beyond teaching [students] to read and write in school, they need to help them learn and use aspects of language associated with the academic discourse of the various school subjects. They need to help them become more aware of how language functions in various modes of communication across the curriculum. They need to understand how language works well enough to select materials that will help expand their students' linguistic horizons and to plan instructional activities that give students opportunities to use the new forms and modes of expression to which they are being exposed. Teachers need to understand how to design the classroom language environment so as to optimize language and literacy learning and to avoid linguistic obstacles to content area learning (Wong Fillmore & Snow, 2000:7).

ESOL teaching is a profession with a high turnover of employees and the different qualification requirements around the world mean that not all teachers have had the same training experience to prepare them to teach English as a second language.

"In my first few jobs as an ESOL teacher, I struggled with teaching English grammar to a class that knew much more about it than me. I would tell my students that because I had grown up speaking English, I hadn't been made to learn the grammar." - *Teacher 6*.

By using the revised FIACS, teachers have a record of their lessons and are able to identify where they can give, and are giving, students opportunities to express the target language of the lesson. By doing this, they will be optimising the language environment in the classroom and giving their students greater ability to practise language use.

When learning anything, the support from teachers is vital to successful attrition. In ESOL, students can struggle because of the culture and language barrier. Lack of student interaction might be misconstrued as ESOL students not being motivated to learn. In fact, most ESOL students desire success just as any other students but may not feel confident in participating in lessons due to these barriers.

"I was actually surprised when I looked back at the transcripts, how much some of my students actually talked. Some of my students are usually so quiet that I didn't know they knew as much as they did. But when I asked them more questions, they really came out of their shells."

- Teacher 4.

"Sometimes it feels daunting as a teacher when you're trying to make sure that you're not dominating the lesson and you're giving the students a chance to speak. Being a part of this exercise has really made me more aware of how much I talk in class and also helped me to identify ways in which I could give my students more opportunities to practice language."

It is important for ESOL students to practise interacting in English in order to develop their spoken communication. Depending on their English language abilities and their previous language learning experiences, learning to converse in English may be intimidating and frustrating, but students need to engage in interaction to develop their skills and to enable the teacher to identify their strengths and weaknesses in class.

⁻ Teacher 8.

Many ESOL teachers will become preoccupied with the negatives and overlook of the positives of their teaching. By analysing their interaction they begin to analyse our speech and thought process also. When teachers discuss their own interaction, they can view their lessons from an outside perspective and gain a more objective understanding of their lessons and their students' language development.

Although the new revised FIACS is not a tool with which to evaluate teaching practices, its use can help teachers to become more self-aware of their own teaching behaviour. Not much is known about teaching-group activities in which some form of coding behaviour is conducted amongst the teachers. What follows, then, can only be considered speculation:

Interaction Analysis in Micro-teaching

Micro-teaching is based on the assumption that there are certain behaviour strategies which are key to effective teaching. By concentrating on these strategies in a teachertraining program, it should be possible to improve teacher performance.

A micro-teaching program is designed to expose the trainee to an organised curriculum of miniature teaching 'encounters'. At each step along the way, a teaching strategy is discussed until it can be incorporated into the short teaching session. The trainee then teaches the session to a small group of students, under the observation of a trainer and perhaps some peers. After the session, the students are dismissed and the teaching is assessed by those still present. The trainee is given time to think about this assessment and to make modifications to their teaching approach. They then teach again to demonstrate improvement.

Imagine, for a moment, that micro-teaching is performed not only with a trainer observing, but that trainer could code the interaction on a time line display. If the categories and display could be constructed to highlight the features of teaching that were being practised, there would be several advantages. First, a time line display could focus attention on specific points of the lesson. This advantage, for example, might lead the trainee and the trainer to agree that at the 12-minute mark, teacher questions began to stimulate student initiation. Attention could be given to the questions before and after that point. Secondly, coding behaviour removes vagueness inherent in some micro-teaching objectives. For example, 'set context' presumably refers to anything and all things that a teacher can do to introduce a topic to students and interest them in it. There are, however, specific things that a teacher can do to engage the interest of a student in the task at hand. A discussion of these actions would be necessary in order to select appropriate categories for coding. These discussions might help to make micro-teaching a more valuable teacher-training method than without interaction analysis. Advantages would be greater specification of the skill to be practiced and more objective information about the teaching performance itself.

Video playback might be a useful resource for teachers to observe their own teaching but in long periods of teaching, this becomes time-consuming and therefore inefficient. Interaction analysis feedback is much faster and can focus on specific skills providing the behaviour patterns can be identified in the display format. It is quite possible that interaction analysis combined with micro-teaching would provide a useful training tool to develop teacher-behaviour. This type of reflection upon teaching behaviour enables teachers to be better able to monitor, make real-time decisions and respond to the changing needs of their students than teachers who are less reflective (Yost et al., 2000; McMeninam et al., 2003 as cited in Mann, 2005). However, teachers need support in structuring this evaluation (Copland, Ma and Mann, 2009). The revised FIACS provides a clear framework for teachers to see the interaction that has taken place in the classroom and therefore can increase the teacher's awareness of interaction in the classroom.

Some Advantages of Using Interaction Analysis Category Systems in Micro-teaching:

- Teaching objectives and live students add to the realism of the practice.
- Total class responsibility is avoided during practice.
- Step-by-step improvement is assessed through the trainee being given an opportunity to teach again after receiving feedback on the previous lesson.
- Reinforcement after a successful use of a teaching strategy is almost immediately apparent.
- Practice can move from the simple to the more complex skills.
- It is possible for the students to evaluate their teachers.

Some Disadvantages

- Equipment failure, if using video or audio recording, can be troublesome.
- Analysis might not be appropriately conceptualised before feedback.
- Organisation of getting live students and equipment requires thorough planning

and beforehand.

• Micro-teaching is not appropriate for long periods of teaching.

6.3 Conclusion

This study has contributed to both the studies of classroom interaction and ESOL by exploring the actual practices of the learners in this setting. The results of this study indicate that students and teachers in an ESOL class may have different interactional preferences for learning than those of non-ESOL teachers and students and in order to account for these differences, a new and more ESOL-specific category system has been developed.

The use of an interaction analysis category system in teacher training may be able to help teachers become more self-aware of their own teaching behaviour. This could be deployed in a micro-teaching context, where trainees could have the opportunity to gain a more objective insight into their own and their colleagues teaching performances.

A discussion of the implications these findings have for practical implementation of FL learning, the limitations of the current study, and suggestions for future research will be discussed further in the next chapter.

CHAPTER 7: CONCLUSIONS

7.1 Introduction

This chapter first summarises the findings of the study and follows with some general pedagogical implications based on the research, recommendations for instructional design of ESOL courses, the limitations of the study, and finally offers suggestions for further research.

7.2 Pedagogical Implications

The following sections examine the pedagogical implications of the findings outlined in the previous chapters. The results will be discussed in terms of the nature of the task rather than by order of the research questions as there are many overlaps between learner practices, course completion and linguistic outcomes.

Interactional Partner

In this study, participants completed far more Student-Teacher tasks than Student-Student tasks, but neither type of task was a significant predictor of interaction. Previous studies (Fredrickson, et al., 2000; LaPointe & Gunawardena, 2004; Jiang & Ting, 2000) had mixed results as to which interaction partner was preferred by students and whether it was more beneficial to for the teacher to speak to students or to just let the students talk amongst themselves.

From what is currently understood about the role of interaction in language learning, Foreign Language learners should be encouraged to communicate regardless of the partner with whom they are communicating. Additionally, this would suggest that the use of a language tutor, someone who is not fluent in the language but above the current level of the student, might also provide an extra resource with whom a student can communicate and equally as useful as fellow classmates or the teacher in terms of language learning. Twigg (2002) emphasises that this knowledge could be of particular interest to language schools because some promote student-student interaction as a way to "save expensive faculty time". In a more altruistic sense, encouraging students to work with a language tutor also makes sense in a Vygotskyan approach to language learning.

Interactional Purpose

Based on the finding that language and cultural tasks together promote language development, FL practice together with reflection and discussion of cultural and linguistic topics should be encouraged. As English is a commonly taught language, there exist many resources available. Given this situation, it may be helpful for students to exchange ideas on resources and to get teacher feedback on other resources, in print or electronic form.

Students in this study completed more receptive or productive tasks than individual

tasks. This was likely because they were more conducive to interaction and provided collaborative learning opportunities. Although interactional tasks focus on the students developing their language skills through practice rather than focusing on discrete language learning frameworks, it is shown that students enjoy individual activities even though they do not allow as many opportunities for the students to take an active part in the lesson, as opposed to sitting and being lectured at. Meunier (1998) points out that students initially may harbour anxiety towards having to communicate in the target language. This offers a possible explanation as to why students may prefer individual tasks to collaborative tasks: they feel the need to complete a task that more closely resembles something familiar. Perhaps given that the majority of the participants in this study were of Chinese backgrounds, where student interaction in lessons is not as frequent, it may take some students more than eight weeks (the duration of the course) to develop a comfort level with collaborative based learning.

One specific recommendation is to create tasks which scaffold task types: tasks which facilitate the development of communicative competence, but are still individual in nature as a way to 'breach the gap' and introduce students to a more communicative style of learning whilst not obviously straying too far from their previous language learning experiences. One example of how this could be done would be a gap fill exercise, where students have to fill in the appropriate words or phrases that are missing in a text or conversation transcript. This may be beneficial, too, in preparing students for a conversation practice, either linguistically or simply in order to make them feel more psychologically prepared.

Although collaborative tasks should be emphasised as a method to encourage

interaction in the classroom, it is possible that receptive tasks may help students stay on track by providing an environment with which students can compare themselves with their classmates. Students were able to listen to various taped recordings of conversations, for example (a receptive task) and would likely benefit from and be interested in archived audio and video sessions. These could be student-student (at the novice level, or more advanced students), or student-teacher, or with native speakers. This would provide more variety for the receptive tasks.

Possibly the most significant pedagogical implication in interaction in the classroom is that all task types are valuable to students' learning, regardless of whether they involve interaction or not. Students in this setting completed a wide variety of collaborative and individual tasks, as noted by Twigg (2002) as being essential for successful FL courses.

As previously discussed, this eight week course may not have been a long enough time for students to experience complete comfort with the course and their teachers, relating to Meunier's (1998) finding of students' initial feelings of anxiety. Based on this possibility, the implications for interaction may be similar for collaborative tasks versus non-collaborative tasks. Specifically, scaffolding of task types, in addition to scaffolding of language learning tasks, might prove to be a way to help students feel more comfortable with learning methods that they have little experience or comfort with. An eight week course may not be long enough for a student to establish complete comfort with the learning environment, so by scaffolding the tasks, students could be exposed to additional opportunities to interact in the classroom. In conclusion of the pedagogical implication section, curriculum coordinators and teachers need to be aware that the facilitation of interactional tasks and giving students opportunities to interact are necessary for students and that based on the previous assumptions of learning, learning needs, or autonomy level of the students, each student will come to the class with a different set of preference and expectations for learning. Active participation and interaction of students, the completion of assignments and completion of language learning tasks have been shown to be possible indicators of course completion and language development, and these are the three things that should be emphasised in future FL courses with FL students (Lie, 2008).

Language Learners

In addition to the implications of this study that are more directly related to the research questions, it is important to again point out that the students and teachers in this study may interact differently than students and teachers in other contexts. This is because of the teachers' training background on the CELTA course, which emphasises interaction and student-talk opportunities. Additionally, most of the students in this study were Chinese. Interaction between students and teachers in the classroom is limited in most Chinese schools and so these students may have been more reticent to interact than students of other nationalities. However, the students in this study may have been very highly motivated to participate in the course because of the limited opportunities for free English language evening classes in their community.

7.3 General Recommendations for ESOL Courses

Amidon & Simon (1965) made an attempt to determine the extent to which interaction analysis had been used in teacher education programmes. They concluded the impact of interaction analysis on students and teachers had begun to be felt at that time:

"Student teachers feel that Interaction Analysis is significant because it helps make operational much of what they have already learnt about educational methods and theory. Students also appear to think they have gained insight into their teaching behaviour and that this insight into their teaching behaviour and that this insight will make it possible for them to adjust their behaviour to various types of teaching situations." (1965:88)

Adult learners should be in an environment that provides support and guidance, but that ultimately allows individuals to make choices in how learning happens. This recommendation is in line with Knowles's (1998) and ragogic learning principles which state that are six assumptions related to the motivation of adult learning:

- Need to know: Adult learners need to know the reason that they are learning something.
- 2. **Foundation:** Previous language learning experience (including error) provides the basis for learning activities.
- 3. **Self-concept:** Adult learners need to be responsible for their own decisions on education; involvement in the planning and evaluation of their instruction.
- 4. Readiness: Adult learners are most interested in learning subjects having

immediate relevance to their work and/or personal lives.

- 5. **Orientation:** Adult learning is problem-centred rather than content-oriented.
- 6. **Motivation:** Adult learners respond better to internal versus external motivators.

Based on the findings of this study, the following recommendations are given:

- Give explicit guidance as to exactly what topics will be covered the classes and what students will be required to already know themselves, or in collaboration with an outside source. This will help to make sure that students do not feel disadvantaged in classroom interactions by lack of knowledge compared to their classmates.
- Allow activities that encourage students to interact verbally, both with the teacher and with each other. While this study revolves around what interaction types occur between teachers and students, most of the focus seems to be on examining mainly student experiences in this context. It should be noted that the most challenging aspect for the teachers was the planning of interactive tasks in this study because of the fact that it was not possible to guarantee the number of students for each session. Planning individual tasks was easier as there was a guarantee that they could be achieved even when student numbers per session were low, although several students commented that they enjoyed having the opportunity to host discussions and to communicate with native-speaking teachers about aspects of English culture, as well as language.

There are many opportunities for CELTA trainers and trainees to incorporate and utilise the revised FIACS in a CELTA programme. CELTA trainees could film each others' lessons, watch them back, transcribe their classroom interaction and evaluate the interaction. The strengths and weaknesses of the particular class would be highlighted and explored in a following feedback session. This activity practises self-evaluation, constructive criticism and critical thinking skills and this is very useful for learning. This technique could be used as preparation for assessed teaching practices on the CELTA course.

CELTA trainees can also create and upload videos of their lessons to the Helix Media Library to allow their colleagues to assess the interaction of their lessons. They can also assess the trainees against the CELTA criteria, and this could also help the trainees to pay more detailed attention to their students' language than they might have been able to do in class. This makes it possible to have different people assessing both the trainee's performance as a teacher, and the students' performances as language learners. The trainee can use this tool to provide self-assessment, they can conduct peer-assessment with their colleagues, or the CELTA trainers themselves can use this as another form of assessment for the trainee. They can compare their assessments together, and discuss where and why they digress. CELTA trainees could use the selfassessment methods if they wish to gain an overview of their skills. Furthermore, trainers and trainees can break the evaluation down, for example, if the class is preforming a specific activity like giving a presentation, then it is possible to take different aspects of the FIACS and only assess the types of interaction which will be relevant. An advantage of this method, is that trainees can use them to assess themselves, but they can also be used for peer-assessment to let the other trainees in the group assess a certain teaching practice session.

7.4 Limitations of the Study

One limitation of this study is that the results rely on a self-assessed means of establishing technical comfort level and perceived language level. Even though the study called for students at a pre-intermediate level of English, based on my interactions with participants in this study, I felt that many of them underestimated their language ability.

It did not seem to the teachers that a number of students actually were receiving English tuition from other sources outside of this study, which translated into the teachers feeling as though they needed to provide a lot of assistance in the classroom, obviously affecting interaction levels in the classroom. Many large institutions offering English language courses have help centres where students can get assistance. These help centres frequently have trained staff ready and able to assist with a wide variety of questions. The community orientated approach to this course could not offer the support needed by some students that would have been provided if they had been enrolled in a course at a larger instruction.

Course Participants

Another significant limitation of this study was the sample, both in terms of its small

size and that it was not randomly sampled. As with many studies involving observation based methods of data collection, the sample size of the study can be problematic. And, although there were initially 49 participants in this study, 3 of these withdrew their results which decreased the sample size to 46. The sample used in this study was selected largely opportunistically, because of this the sample lacks diversity which may have a bearing on the results. As mentioned previously, the students come from majority Chinese nationalities, where students are less accustomed to talking in the classroom, so even though the percentage of initiative student talk was higher than anticipated by previous research, it may well have been higher with a sample of students of nationalities more familiar with collaborative learning styles. The teaching sample also, were all selected from recent graduates of the CELTA course who had also been studying for a degree in Linguistics at the same university in the UK, and so the sample of teachers also lacked diversity. It may very well be that there are certain features of interaction prevalent between teachers and students in ESOL which did not occur within this specific sample demographic.

Study Design

Based on the objectives and rationale of this study, a descriptive and then quasiexperimental design was deemed to be the most appropriate. This design captures how FL students and teachers behave in an ESOL course, and then further interprets their behaviour. The current study is exploratory, following the most recent recommendations in the field of English as a foreign language where there are limited empirical studies available. This was essential to further the current knowledge of student-teacher interaction from perceptions to actual practices. The design's weakness is that it does not examine perceptions of learning, thus it does not capture possible affective variables. It is also not within the scope of this study to determine whether or not interaction nor any specific type or function of interaction has any significant bearing on students levels of achievement or assessment grades and so any information relating to the possible effects of interactions on these factors was not assessed.

7.5 Recommendations for Future Research

Although this study produced interesting results about the way in which FL students and teachers interact with each other in an ESOL classroom, these results need to be considered as preliminary. There have been so few studies carried out on the actual interactional practices of ESOL students and teachers, that a number of questions remain unanswered.

Firstly, the results of this current study should be viewed within the context of the population presented: pre-intermediate students of a largely Chinese background learning English with the same university in the UK, being taught by mainly female teachers who were all recent graduates of the CELTA course and all studying for an undergraduate degree in English Language and Linguistics at the same University in the north of England. Similar results might not have been obtained from students and teachers of other demographics, students learning at a different language level or in a different classroom context.

An interesting avenue for future research could be to explore synchronous versus

asynchronous interaction. Online foreign language tutoring is growing in popularity due to its convenience and adaptability. Future research could be conducted into students' preference for computer mediated communication and interaction within this media. The exploration for how this new media may affect classroom interaction, and possibly how current preferences for classroom interaction can have an impact on whether or not online tutoring is a viable method for education, in terms of student achievement. The application of new technology such as this could affect student and teacher satisfaction, in addition to course completion rates, student outcomes and motivation of the students. This could also lead to the further question of whether or not interaction or specific types of interaction between teachers and students in the classroom is really essential for students' learning. This study assumes that interaction in the ESOL is beneficial for student outcomes, but it would be very interesting for further research to attempt to test this. In a similar vein, it would also be interesting to examine why classroom interaction is emphasised as highly beneficial in teaching methodology and academia, in light of research that shows that there is in fact no statistically significant difference between levels of interaction and final grade outcomes in language courses. Why does the majority of the research into ESOL teaching, including this study, harbour under the assumption that interaction matters at all? These questions have become apparent to me whilst in the course of analysing and evaluating the field of this study. Though it is not within the scope of this study to answer these questions, it is important for researcher to be critical of the existing body of the work in their field and so this is an area which needs to be examined.

Another interesting avenue for future research would be to determine whether or not, and if so to what extent, individual differences or perceptions on how language is learnt have an effect on student-teacher interaction. Do participants whose goal was to obtain a high degree of communicative competence make more of a conscious effort to interact verbally to a greater amount in class? For those students who come to class with a belief that grammar translation type activities lead to language acquisition, does communicating through verbal interaction change their views of language learning?

Along those same lines, how do student readiness variables (for example, motivation, previous language learning experiences and expectations) affect both the learning outcomes and course completion and how do these variables affect how an individual interacts? Is it a domino effect? There has been some preliminary research into individual differences related to student satisfaction of courses, but this area needs to be investigated further.

Another area that would be interesting to conduct further research into is the role receptive tasks play in linguistic development. For example, do students first seek out linguistic models of the task to read or listen to before completing a productive task? Do students find these useful, specifically at the very beginning stages of language learning?

The benefit that participants felt they received from completing tasks is another potential area for future research. Were the tasks helpful for course completion and the development of language skills because they offered the participants the chance to interact? Or did the completion of these tasks act as a motivator, having provided the opportunity for interaction in a secure and relatively non-threatening environment?

Additionally, it would be valuable to gain in depth understanding of what a small group of foreign language learners interact in this setting, how their differing learning variables possibly affect their interactional preferences and how they react to a new learning environment.

A final question is do students prefer to complete more interactional tasks or noninteractional tasks? Non-interactional tasks, such as writing, may be appealing to students as they are more private, because they are less face threatening, and because they may possibly resemble some students' previous language learning experience. Do students feel differently about the value of interaction based tasks on their language learning after the completion of a course than at the beginning of the course?

7.6 Conclusion

As English has become and is still becoming the language of expansion throughout most of the world, a lingua franca among foreign language speakers, the demand for English Language teaching has risen. ESOL is a rapidly expanding industry with features and preferences for interaction in the classroom that differ from other kinds of teaching. It is useful to examine practices that can positively affect students' experiences in ESOL courses. This study has contributed to both the fields of Applied Linguistics and TESOL by exploring actual practices of students and teachers in this setting. The results of this study indicate that participants have different interactional preferences for second language learning than in native language contexts. This study has contributed to both the studies of classroom interaction and ESOL by exploring the actual practices of the learners in this setting. The results of this study indicate that students and teachers in an ESOL class may have different interactional preferences for learning than those of non-ESOL teachers and students and in order to account for these differences, a new and more ESOL-specific category system has been developed in order to provide an accurate reflection of interaction practices and what is considered a significant interactional finding in this context.

Very few studies have examined actual interaction practices in the fields ESOL and FL interaction, instead drawing assumptions on interaction practice in ESOL through analysis of non-ESOL contexts. As this is a relatively new field of research, much remains to be investigated to further our understanding of the unique challenges students and teachers face in this environment and how learner preferences and needs are best met.

REFERENCES

Abdolrahimi Javed, S., Zahed- Babelan, A. & Namvar, Y. (2013). "A Study on the State of Teacher-Student Verbal Interactions during Teaching Process and Its Relationship with Academic Achievement in Ardabil". *International Research Journal of Applied and Basic Sciences, 4* (7): 1909-1913.

Adegoke, B.A., & Emeke, E.A. (2008). "Measuring Social Forces in the Classroom by Flanders' Technique". In M. Boucouvalis and R. Aderinoye (Eds.) *Education for Millennium Development*, Ibadan: Spectrum Publishers, pp. 208 – 215.

Ahmad, F. & Aziz, J. (2009). "Student's Perceptions of the Teacher's Teaching of Literature Communicating and Understanding Through the Eyes of the Audience." *European Journal of Social Sciences*, *7* (3), 17-39.

Amidon, E. (1967). "Interaction Analysis and Its Application to Student Teaching." *Theoretical Bases for Professional Laboratory Experiences in Teacher Education*,
Forty-Fourth Yearbook of the Association for Student Teaching , Dubuque, Iowa: Wm.
C. Brown, Co.

Amidon, E. & Hough, J. (1967). *Interaction Analysis: Theory, Research and Application*. Palo Alto: Addison-Wesley.

Amidon, E. & Hunter, E. (1967). *Improving Teaching: The Analysis of Classroom Verbal Interaction*. New York: Holt, Rinehart and Winston.

Anorue, C. (2004). *Patterns of Student Interaction in Social Studies in Imo State Secondary Schools.* Ph.D. Thesis. Nigeria: University of Port Harcourt.

Atakins, A. & Brown, H. (2001). *Sinclair & Couthard's "IRF" model in one-to-one classroom: an analysis*. Available From: <u>http://www.jstor.org/pss/1170088</u>. [Last accessed: 2.03.2016].

Bandura, A. (1986). Social Foundations of Thoughts and Action: A Social Cognitive Theory. Englewood Cliffs, NJ: Prentice Hall.

Barnes, D. & Todd, F. (1977). *Communication and Learning in Small Groups*. London:Routledge.

Barnes, D. & Todd, F. (1995). *Communication and learning revisited, making meaning through talk*. Portsmouth, NH: Boynton/Cook Publishers Heinemann.

Beaudoin, M. (2002). "Learning or Lurking? Tracking the "Invisible" Students." *The Internet and Higher Education, 5,* pp 147-155.

Benati, A.G. (2013) Issues in Second Language Teaching. Sheffield: Equinox.

Blatt, B., Confessore, S., Kallenberg, G., Greenberg L. (2008). "Verbal Interaction Analysis: Viewing Feedback through a Different Lens". *Journal of Teaching and Learning*, 20 (4), 329-333. Boyd, R. D. & DeVault, M. V. (1966) "The Observation and Recording of Behaviour." *Review of Educational Research, 36,* 529-551.

Brower, E.W., Dejonge, J.O. & Stout, V.J. (2001). *Moving to Online: Making the Transition from Traditional Instruction and Communication Strategies*. London: Corwin Press.

Brown, H. (1994). *Teaching by Principles: An Interactive Approach to Language Pedagogy*. Prentice-Hall Regents: Englewood Cliffs, NJ.

Cardelle, M. & Corno, L. (1981). "Effects on Second Language Learning of Variation in Written Feedback on Homework Assignments." *TESOL Quarterly*, *15* (3), 251-261.

Carr, S. (2000). "As Distance Education Comes of Age, the Challenge is Keeping the Students." *Chronicle of Higher Education*, n.p.

Carr, S., Ledwith, F. (2000). "Helping Disadvantaged Students." *Teaching at a Distance, 18,* 77-85.

Carroll, S. (2001). *Input and Evidence: The raw material of second language acquisition*. Amsterdam: John Benjamins B. V.

Cenoz, J., Hornberger, N. H. (2008). "Classroom Discourse: Approaches and Perspective of Language and Education." In: *Encyclopedia of Language and Education: Knowledge about Language. Volume 6.* (2nd.Ed). California: Springer.

Chapelle, C. (1997). "CALL in the year 2000: Still in search of research paradigms?" *Language Learning and Technology Journal, 1* (1), 19-43.

Chaudron, C. (1988). Second Language classroom: Research on Teaching and Learning. Cambridge: Cambridge University Press.

Christie, F. (2002). *Classroom Discourse Analysis: A Functional Perspective*. London: Continum.

Cole, P. G. & Chan L. K. (1994). *Teaching Principles and Practices (2nd. Ed.)*. New York: Prentice Hall.

Copland, F., Ma, G., & Mann, S. (2009). "Reflecting in and on post-observation feedback in initial teacher training on certificate courses." *English Language Teacher Education and Development, 12,* 14-22.

Curtis, S., Duchastel, J. and Radic, N. (1999). "Proposal for an Online Course." *ReCALL*, *11* (2), 38.45.

Dankin, M. J. and Biddle, B. J. (1974). *The Study of Teaching*. New York: Holt, Rinehart & Winston.

Darling Hammond, L., LaPointe, M., Meyerson, D. & Orr, M. (2007). *Preparing School Leaders for a Changing World: Lessons from Exemplary Leadership Development Programs*. Stanford: The Finance Project.

Diaz, D. & Cartnal, R. (June/July, 2006). "Term Length as an Indicator of Attrition in Online Learning. Innovative. Available from: <u>http://www.innovateonline.info/index.php?view=article&id=196</u> [Last accessed: 14.03.2016].

De Bot, K., Lowie, W., & Verspoor, M. (2005). *Second Language Acquisition: An Advanced Resource Book.* London: Routledge.

Eber, M. & Holzinger, A. (2005). "Lurking: An Underestimated Phenomenon." *IEEE MultiMedia*, *12* (4), 70-75.

Edwards, J. (2005) "Exploratory talk in peer groups: exploring the zone of proximal development." In *Fourth Congress of the European Society for Research in Mathematics Education. Spain: Sant Feliu de Guíxols.*

Fathi-Azar, E. (2003). Teaching Methods and Skills. Tabriz: Tabriz University

Flanders, N.A. (1970) Analysing Teacher Behaviour. Reading: Addison Wesley.

Furst, N. (1967). "The Effects of Training in Interaction Analysis on the Beaviour of Student Teachers in Secondary School." In Amidon, E. & Hough, J. (Eds.), *Interaction Analysis: Theory, Research and Application*. Palo Alto: Addison-Wesley.

Jackson, P.W. (1990) Life in Classrooms. New York: Teacher College Press.

Fredericksen, E., Pickett, A., Shea, P., Pelz, W. & Swan, K. (2000). "Student Satisfaction and Perceived Learning with Online Courses: Principles and Examples from SUNY Learning Network." *Journal of Asynchronous Learning Networks, 4* (2).

Freeman, J. & Anderson, M. (2007). *Lurking: A New Learning Style?* Paper presented at the TXDLA conference, Galveston Island, TX.

Gage, N.L. & Berliner, D.C. (1998). *Educational Psychology: A Study Guide*. (6th Edn.) Cengage Learning.

Gardner, R., & Lambert, W. (1972). *Attitudes and Motivation in Second Language Learning*. Rowley, MA: Newbury House.

Gass, S. M. (2003). "Input and Interaction." In Doughty, C. & Long M. H. (Eds.),

Handbook of second language acquisition (pp. 224-255). New York: Basil Blackwell.

Gibbons, P. (2003). "Mediating Language Learning: Teacher Interactions with ESL Students and a content based classroom." *TESOL Quarterly*, *37* (2), 247-273.

Gross, B. (1993). Tools for Teaching. New Jersey: John Wiley and Sons.

Gutierrez, J. (2000). "Instructor-student Interaction." *USDLA Journal*, 14 (3). Available From: http://www.usdla.org/html/journal/MAR00_Issue/Instructorstudent.htm [Last Accessed 14/12/2015].

Haalen, T., & Miller, G. (1994) *Interactivity as a Predictor of Student Success in Satellite Learning Programs*. DEOSNEWS, 4 (6) LISTSERV@ PSUVM.PSU.EDU.

Hafiz- Mahmud, I. *et al.* (2008). "Teacher-Student Verbal Interaction Patterns at the Tertiary Level of Education." *Contemporary Issues Educ. Res, 1., 1* (1), 45-50.

Hall, J. (1993). "The Role of Oral Practices in the Accomplishment of Our Everyday Lives: The Sociocultural Dimension of interaction with Implications for The Learning of Another Language." *Applied Linguistics* 14, 145-166.

Hai, S.K. & Bee, L. S. (2006). "Effectiveness of Interaction Analysis Feedback on the Verbal Behaviour of School Teachers." *The Asia Pacific Journal of Educators and* Hao, Y. (2004). "Student's Attitudes Towards Interaction." *Dissertation Abstracts International*, 65 (10). 3768. (UMI No. AAT 3150582).

Hatch, E. (1978). "Discourse Analysis and Second Language Acquisition." In Hatch, E.(Ed.), *Second Language Acquisition: A Book of Readings*. Rowley, MA: Newbury House.

Jiang, M., & Ting, E. (1998). *Course Design, Instruction, and Student's Behaviours: A Study of instructional Variables and Student's Perceptions of Online Learning.* Paper presented at the American Educational Research Association, April. San Diego, CA.

Jin, L., & Cortazzi, M. (1998). "The culture the learner brings: A bridge or a barrier? InM. Byram & M. Fleming (Eds.), Language learning in intercultural perspective:Approaches through drama and ethnography. Cambridge, England: CambridgeUniversity Press.

Kearsley, G. (1995). "The Nature of Interaction in Learning." In Koble, M. (Ed.) *The American Center for the Study of Distance Education Invitational Research Conference in Distance Education: Toward Excellence in Education*. University Park, PA: Penn State University Press. Kearsley, G. (1999). Exploration in Learning & Instruction: The Theory into Practice Database. Available from: http://tip.psychology.org/theories.htm [Last Accessed: 22.06.2016].

Knowles, M., Elwood, F., & Swanson, R. (1998). *The Adult Learner*. Houston: Gulf Publishing.

Krashen, S. (1985). The Input Hypothesis. London: Longman.

Lambert, P.; Goodwin, L. & Roberts R. F. (1965). "A Note on the Use of Flanders Interaction Analysis." *Journal of Educational Research 58*, 222-224.

Lambert, R. (1991). "Education and Foreign Languages." *Occasional Papers*. Washington: National Foreign Languages Centre.

Lantolf, J. (2000). "Introducing Sociocultural Theory." In Lantolf, J. (Ed.), *Sociocultural Theory and Second Language Learning* (pp.1-26.) New York: Oxford University Press.

Lantolf, J. (2001). *Sociocultural theory and second language learning*. Oxford: Oxford University Press.

Lie, K. (2008). Virtual Communication: An Investigation of Foreign Language

Interaction in a Distance Education Course in Norwegian. Unpublished Doctoral Dissertation, University of Texas, Austin.

Long, M. (1980). *Input, Interaction, and Second Language Acquisition*. Unpublished doctoral dissertation, University of California, Los Angeles.

Long, M. (1983). "Native Speaker/Non-Native Speaker Conversation and The Negotiation of Comprehensible Input." *Applied Linguistics, 4,* 126-141.

Luk, J. C. M. (2004). "The Dynamics of Classroom Small Talk". *Issues in Applied Linguistics, 14* (2), 96-110.

Mann, S. (2005). "The language teacher's development." *Language Teaching. 38* (3), 103-118.

McKay, Sharon & Schaetzel, Kirsten, *Facilitating Adult Learner Interactions to Build Listening and Speaking Skills*, CAELA Network Briefs, CAELA and Center for Applied Linguistics, July 2008

Medley, D. M., & Mitzel, H. E. (1958) "A Technique for Measuring Classroom Behaviour," *Journal of Educational Psychology 49*, 86-92.

Meunier, L. (1998). "Personality and Motivational Factors in FL Communication." In Muyskens, J. (Ed.), *New Ways of Learning and Teaching: Focus on Technology and* Foreign Language Education. (pp. 145-198). Boston: Heinle.

Moore, M. & Kearsley, G. (1996). *Education: A Systems View.* Wadsworth: Belmont, CA.

Mori, J. (2007). "Border Crossings? Exploring the Intersection of Second Language Acquisition, Conversation Analysis, and Foreign Language Pedagogy." *The Modern Language Journal*, *91* (1), 849-862.

Needles, M. C. (1988). "A New Design for Process: Product Research on the Quality of Discourse in Teaching." *American Education Research Journal*, *25* (1), 503-526.

Nobuyoshi, J. & Ellis, R. (1993). "Focused Communication Tasks and Second Language Acquisition." *ELT Journal*, *47*, 203-10.

Northrup, P. (2002). "Online Learners preferences for Interaction." *Quarterly Review* of Distance Education, 32, 219-226.

Nunan, D. (1991). Language *Teaching Methodology: A Textbook for Teachers*. New Jersey: Prentice Hall.

Odundo, P. (2013). "Effects of Application of Instructional Methods on Learner Achievement in Business Studies in Secondary Schools in Kenya." *International Journal of Education and Research, 1* (5) 1-22. Oranu, P. C. (2012). "The Extent of Student's Initiation of Ideas in The Classroom". *The International Journal of Academic Research in Business and Social Sciences, 2* (1), 293-300.

Parker, A. (1999). "Interaction in Education: The Critical Conversation." *Educational Technology Review*, *12*, 13-16.

Paulus, T. (2007). "CMC Models for Learning Tasks at a Distance." *Journal of Computer-Mediated Communication*, *12* (4), 9.

Pica, T. (1987). "Second Language Acquisition, Social Interaction, and The Classroom." *Applied Linguistics*, 8, 3-21.

Pontecorvo, C. (1993). "Social Interaction in the Acquisition of Knowledge". *Educational Psychology Review Journal*, 5 (3), 293-310.

Putz, M. & Aertselaer, N.V.J. (2008). *Deconstructing Contrastive Pragmatics: Interlanguage and Cross-Cultural Perpsectives*. Berlin: Walter De Grutyer.

Relph, T. L. (2015) *Categorising Interaction in Teaching English to Speakers of Other Languages.* Unpublished Dissertation.

Restauri, S. (2006). Faculty-Student Interaction Components in Online Education: What are the Effects on Student Satisfaction and Academic Outcomes? Unpublished doctoral dissertation, Capella University.

Reynolds, D. & Peter, C. (2009). *School Effectiveness Research, Policy and Practice Over Three Decades*. London: Routledge.

Rivers, W. M. (2000). *Interactive Language Teaching*. Bejing: People's Education Press.

Roblyer, M. & Ekhmal, L. (2000). *How Interactive Are Your Courses? A Rubic For Assessing Interaction*. Paper presented at the Distance Learning Administration 2000 Conference, Callaway, GA.

Rosenshine, B. V. (1971). *Teaching Behaviours and Student Achievement*. London: National Foundation for Education Research in England and Wales.

Saba, F. (2007). "Postmodern Theory of Distance Education, Distance Education Systems of the Future." *Journal of Education Technology Systems*, *17* (1), 215-250.

Sabry, K. & Baldwin, L. (2003). "Web-based Learning Interaction and Learning Styles." *British Journal of Educational Technology, 34* (4), 443-54.

Safari, S. (2010). "Resource Managers: the Power to Review Its relationship with Organisational Climate From the Perspective of School Teachers". *Journal of Industrial Psychology. 1* (3).

Sahlberg, P. (2007). "Education Policies for Raising Student Learning." *Journal of Education Policy. 22* (2), 147-171.

Sahlberg, P. (2008). "The more you talk, the more you learn: Missing Conditions for cooperative Learning in secondary Schools." *European Journal of Education*. *43* (2): 151-172.

Schultz, K. (2010). *Listening to Sounds of Silence in the Classroom*. Available from: http://www.store.tcpress.com/0807750174s.html [Last accessed 2.03.2016].

Seedhouse, P. (2004). "Conversation Analysis Methodology." *Language Learning, 54* (51) 1-54.

Seedhouse, P. (2005). "Conversation Analysis and Language Learning". *Language Teaching*, *38* (4), 165-187.

Sharan Shahi, R. (2010). Observation and Analysis of Classroom Teaching at Tertiary Level. *Bilingual Journal of Humanities and Social Sciences*. *1* (1), 1-4.

Simpson, M. & Tuson, J. (2003). "Using Observations in Small-Scale Research." Glasgow: University of Glasgow Press.

Smith, B. & Othanel B. (1968). "Recent Research on Teaching: An Interpretation." *The Education Digest 33*, 51-51.

Stokoe, E. (2011). Simulated Interaction and Communication Skills Training: The 'Conversation Analytic Role-Play Method'. In Antaki, C. (Ed.), *Applied Conversation Analysis: Changing Institutional Practices*. Basingtoke: Palgrave Macmillan.

Swain, M, (1985). "Communicative Competence: Some Roles of Comprehensible
Input and comprehensible Output in Its Development." In Gass, S. & Madden, C.
(Eds.), *Input and Second language Acquisition* (pp. 235-253). Rowley, MA: Newbury
House.

Swain, M. (1995). "Three Functions of Output in Second Language Learning." InCook, G. & Seidlhofer, B. (Eds.), *Principle and Practice in Applied Linguistics:Studies in Honor of H.G. Widdowson* (pp. 125-144). Oxford: Oxford University Press.

Swain, M. (2000). "The Output Hypothesis and Beyond: Mediating Acquisition Through Collaborative Dialogue." In Lantolf, J. (Ed.), *Sociocultural Theory and Second Language Learning* (pp.97-114.) New York: Oxford University Press.

Tello, S. (2002). *An Analysis of The Relationship Between Instructional Interaction and Student Persistence in Education*. Unpublished doctoral dissertation, University of Massachusetts, Lowell.

Terrell, T. (1982). "A Natural Approach." In Blair, R. W. (Ed.), *Innovative Approaches to Language Teaching* (pp. 160-173). Rowley, MA: Newbury House Publishers.

Thurmond, V., Wambach, K., & Connors, H. (2002). "Evaluation of Student

Satisfaction." The American Journal of Distance Education, 16 (3), 169-189.

Twigg, C. (2002). Improving Quality & Reducing Costs: Designs for Effective
Learning Using Information Technology. Observatory Report. Available From:
https://www.luminafoundation.org/files/publications/collegecosts/twigg.pdf [Retrieved:
7/12/2015].

Vygotsky, L. (1978). *Mind and Society: The Development of Higher Psychological Processes.*

Cambridge, MA: Harvard University Press.

Wagner, E. (1994). "In support of a functional definition of interaction." *The American Journal of Distance Education*, 8 (2), 6-29.

Walker, R. (2002). *Time for the Demise of Classroom Research?* Available From: https://www.uea.ac.uk/edu/ddncl/articles/Time_for_the_Demise.pdf [Last Accessed: 27.11.2015].

Walsh, S. (2006). Investigating Classroom Discourse. London: Routledge.

Walsh, S. (2013). *Classroom Discourse and Teacher Development*. Edinburgh:University Press.

Weimer, M. (1993). *Student's Recommendations for Encouraging Participation*. Available From: http://www.facultyfocus.com/articles/effective-classroommanagement/student-recommendations- for-encouraging-participation [Last Accessed: 19.10.2016].

Wiseman, S. & Start, K.B. (1965) "A Follow Up of Teachers Five Years After
Completing Their Training." *British Journal of Educational Psychology*, *35* (3), 342-361.

Wong Fillmore, L. & Snow, C (2000). *What Teachers Need to Know about Language*. McHenry, IL: Delta Systems.

Wragg, E. C. (2002). *An Introduction to Classroom Observation. (2nd. Ed.)*. London: Routledge.

Yang Xueyan. (2003) "On the Study of Foreign Language Teachers' Classroom Strategies". *Journal of Foreign Language Teaching and Research*, 35 (1), 54-61.

Zahed- Babelan, A. & Moenikia, M. (2010). "A Study of student-teacher verbal interaction and its relation with student's educational achievement". *Journal of Education*, *6* (2), 107-122.

APPENDIX I

A sample copy of the consent form completed by the teachers on the Course Y classes.

Teacher Consent Form

Welcome to Course Y free English Lessons!

As you are aware, these free classes give you the opportunity to use your newly qualified teacher status and get some practise. All lessons will be video recorded and the data collected will be uploaded to the YSJ media library where only YSJ students will be able to access this. They will be used for academic and research purposes by your fellow teachers. You will also be able to view your own video from the media library.

- 1. I confirm that I have read and understand the information above and have had the opportunity to ask questions.
- 2. I agree to take part in teaching the free English classes.
- 3. I agree to the classes being audio recorded.
- 4. I agree to the classes being video recorded.
- 2. I give permission for anything I say in the lessons to be guoted in publications. (Your name and personal information will not be published).
- 3. I agree that the recordings will be viewed by my fellow teachers and university staff who will keep them private.
- 4. I agree for the recordings and any relevant information to be used for academic purposes.
- 5. I agree for the recordings and any relevant information to be used for research purposes.
- 6. I agree for the recordings and any relevant information to be used for teachertraining purposes.
- 7. I understand that I can ask to be removed from the data at anytime by contacting the researcher at the email below.

Name of Participant	Date	Signature	
Name of Researcher	Date	Signature	

APPENDIX II

A sample copy of the consent form completed by the students for the Course Y classes.

Student Consent Form

Welcome to the Course Y free English lessons!

These lessons are taught by York St. John University students. All of the student are currently working on the English Language Teaching Project, as part of the English Language and Linguistics Undergraduate degree. All of the students are qualified English teachers.

The lessons will be filmed and the videos used to provide data for your teacher's projects. All videos and any personal information you give to the teachers will be kept private. Only your teachers and York St. John University staff will be able to watch them. The videos will be used for academic and research purposes and may be used to train new teachers in the future.

You can ask to watch the videos yourself, by contacting us using the information below.

- 1. I confirm that I have read and understand the information above and have had the opportunity to ask questions.
- 2. I understand that the lessons are voluntary and that I can stop attending the free English lessons at any time, without giving reason.
- 3. I agree to take part in the free English classes.
- 4. I agree to the classes being audio recorded.
- 5. I agree to the classes being video recorded.
- 6. I give permission for anything I say in the lessons to be quoted in

publications. (Your name and personal information will not be published).

7. I agree that the recordings will be viewed by my teachers and university staff who will keep them private.

8. I agree for the recordings and any relevant information to be used for academic purposes.

9. I agree for the recordings and any relevant information to be used for research purposes.

10. I agree for the recordings and any relevant information to be used for teacher-training purposes.

11. I agree to my written work being photocopied by my teacher.

12. I agree to my photocopied written work being used for academic purposes.

13. I agree for my photocopied written work to be used for research purposes.

14. I agree for my photocopied written work to be used for teacher- training purposes.

15. I understand that I can ask to be removed from the data at any time by contacting the researcher at the email below.

16. I understand that I can request a copy of this consent form by contacting the researcher at the email below.

17. I understand that I can request copies of my teacher's research projects when they have been completed by contacting the researcher at the email below.

Name of Participant Date

Name of Researcher Date

Signature

Signature

You can email the researcher to ask any questions, for further information or to request a copy of this consent form for your personal records:

tiffany.relph@yorksj.ac.uk

APPENDIX III

Sample Consent form sent to participants to use the classroom recordings in this study.

Dear Sir/Madam

I am writing to you because you agreed to take part in the Course Y corpus project in 2014/2015. At that time, you signed a consent form granting the use of video recordings from those classes to be used in further research projects.

I would like to take this opportunity to inform you about a research project I am currently carrying out, and to invite you to participate.

Who am I?

I am studying at York St John University for a Masters by Research in English Language and Linguistics. The title of my current project is 'Categorising Interaction between Teachers and Students in ESOL Classrooms' and as a part of this project, I'm interested in finding out more about what happens in ESOL classes.

How am I trying to find the answers?

To find some answers, I would like to have access to the video recordings made of the classes you attended as part of the Course Y course. I will observe the videos to identify the types of interaction that take place. The aim of the observations is not to make any judgements about the content and delivery of the lesson, or about the conduct of the teacher and students. The observations are simply descriptive and serve the purpose of enabling me to find out how interaction is currently delivered.

Who will benefit from the project?

I strongly believe that I have a responsibility towards the schools and the individuals involved with the project. I will provide feedback to you during the project and, if you wish, will share with you any published findings. I ultimately hope that this research will help to improve people's understanding of how interaction is currently being dealt with in ESOL classes. In the long run, this should benefit both teachers and students.

What will happen if you decide to be involved in the project?

Participation in the project is voluntary. This study is in no way concerned with either teachers' or students' performance. I am very keen that participants feel happy and comfortable with being involved in the research. When I write about what the observations reveal later on, I will not disclose the names of any of the people involved, or what school they attend. If any names of people or schools are mentioned during any of the observed classes, I will change these so that no-one can be identified. I will ensure secure storage of the observation notes by keeping them on password-protected computer files, and any hard copies will be kept in locked filing cabinets which only I will be able to access. No-one except myself will have access to any data relating to the observations. I will primarily write up my research findings in the form of a thesis entitled *Categorising Interaction between Teachers and Students in ESOL Classrooms* which I am happy to share with you. This project has been approved by the York St John University Research Ethics Committee and the authorisation code is 151105.

If you decide that you do not want to be observed, please let me know and I will remove you from any of the observation data. If you agree to participate in the project, but later change your mind, you can withdraw from the project at any time and do not have to provide a reason for withdrawing. If you would like to participate in the project, all you need to do is sign the slip at the end of this letter.

If you would like to discuss this project further with me, then please email me: tiffany.relph@yorksj.ac.uk.

Tiffany Relph Masters by Research in English Language and Linguistics York St John University

Name (please print):

I agree to participate in the research project '*Categorising Interaction* between Teachers and Students in ESOL Classrooms'. I have read and understand my rights as a research participant, as explained in the letter above.

Signed:	
Date:	

APPENDIX IV

Sample transcript of ESOL class observed for this study. Each line represents 3 seconds of interaction with the interaction type coded in the right hand column.

T2- Okay, so everyone	7a
just look at the front? Can anybody	7a
tell me what "he was studying" what tense is that?	4a
All- Past simple T2- Past?	
S2- Continuous T2- past	8
Continuous yeah	3
er, and the next one "he decided?" All- Past	4b
simple T2- Simple, yeah	8, 3
So then, "were travelling"?	4b
All- Past continuous T2- yeah	8, 3
"Went"?	4b
All- Past simple T2- Yeah, yeah, er,	8, 3
"Was raining"? All- Past continuous	4b,8
T2- And "decided" again? All- Past continuous	4b,8
T2- Past simple. Okay S2- yeah!	5, 8
T2- So, there's two different types of	5
verbs there. In the first sentence	5
can anybody tell me if	4a
he was studying first or if he decided first.	4a
All- studying T2- Studying!	8, 3
T2- so, here	5
	1

we are studying (T2 draws timeline on the board)	
and this is the timeline	5
where would decided	4a
go? Would it go before	4a
or after? S8- After T2- After.	4a83
S2- It would go before. T2- Before? S2- yeah.	
So you think he decided, and then he started	3
studying? S7- Er, was. Was.	3,9
I think studying is after the T2- "decided"	9, 5
S2- After he started studying.	8
1.	